

Article



# Neural Stochastic Differential Equations with Neural Processes Family Members for Uncertainty Estimation in Deep Learning

Yongguang Wang \* D and Shuzhen Yao

School of Computer Science and Engineering, Beihang University, Beijing 100191, China; szyao@buaa.edu.cn \* Correspondence: wangyongguang@buaa.edu.cn

Abstract: Existing neural stochastic differential equation models, such as SDE-Net, can quantify the uncertainties of deep neural networks (DNNs) from a dynamical system perspective. SDE-Net is either dominated by its drift net with in-distribution (ID) data to achieve good predictive accuracy, or dominated by its diffusion net with out-of-distribution (OOD) data to generate high diffusion for characterizing model uncertainty. However, it does not consider the general situation in a wider field, such as ID data with noise or high missing rates in practice. In order to effectively deal with noisy ID data for credible uncertainty estimation, we propose a vNPs-SDE model, which firstly applies variants of neural processes (NPs) to deal with the noisy ID data, following which the completed ID data can be processed more effectively by SDE-Net. Experimental results show that the proposed vNPs-SDE model can be implemented with convolutional conditional neural processes (ConvCNPs), which have the property of translation equivariance, and can effectively handle the ID data with missing rates for one-dimensional (1D) regression and two-dimensional (2D) image classification tasks. Alternatively, vNPs-SDE can be implemented with conditional neural processes (CNPs) or attentive neural processes (ANPs), which have the property of permutation invariance, and exceeds vanilla SDE-Net in multidimensional regression tasks.

**Keywords:** deep neural networks; neural stochastic differential equation; neural processes; uncertainty estimates

## 1. Introduction

Deep learning models have achieved great success in many fields, such as image classification [1], computer vision [2], machine translation [3], and reinforcement learning [4]. However, in key fields where safety is at stake, such as in medical diagnoses or autonomous vehicles, the uncertainty estimation of deep learning models is essential for decision making in order to avoid dangerous accidents. Existing studies have shown that deep neural networks (DNNs) models are usually miscalibrated and overconfident in their predictions, which can result in misleading decisions for out-of-distribution (OOD) samples, so it is very important to add credible uncertainty estimates to the predicted values [5].

Bayesian neural networks (BNNs) methods were once regarded as a gold standard for uncertainty estimation in machine learning models [6,7], and the recent benchmark Bayesian method applies a backpropagation-compatible algorithm for learning a probability distribution on the weight of a neural network, which is called Bayes by Backpropagation (BBP) [8]. However, Bayesian methods are very inefficient when performing posterior inference in DNNs with a large number of parameters. On the one hand, in order to improve efficiency, the existing studies adopt the linear subspace feature extracting method of principal component analysis (PCA) in order to construct the parameter subspace of DNNs for a Bayesian inference [9], and the curve parameter subspace method is proposed to build a rich subspace containing diverse, high-performing models [10]. Meanwhile, the latest incremental kernel PCA (InKPCA) approach applies kernel PCA to extract higher



**Citation:** Wang, Y.; Yao, S. Neural Stochastic Differential Equations with Neural Processes Family Members for Uncertainty Estimation in Deep Learning. *Sensors* **2021**, *21*, 3708. https://doi.org/10.3390/s21113708

Academic Editor: Alejandro Linares-Barranco

Received: 15 April 2021 Accepted: 24 May 2021 Published: 26 May 2021

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). order statistical information from DNNs' parameter space, and achieves more accurate results than the PCA and curve subspace inference methods [11]. On the other hand, to approximate the Bayesian inference method, dropout in NNs can be interpreted as an approximation of the Gaussian process (GP), and dropout variational inference (DVI) can be an approximate Bayesian inference approach for large and complex DNN models [12].

Non-Bayesian methods are also studied for uncertainty estimation in DNNs models. For example, the ensemble modeling approach trains several DNNs models with diverse initialization seeds, and uses the predicted values for uncertainty estimation [13]. Meanwhile, if DNNs are trained with a stochastic gradient descent (SGD), the training procedure can average multiple points along the trajectory of the SGD in order to construct a stochastic weight averaging (SWA), which produces much broader optima than an SGD [14]. Due to the dynamics of training DNNs with SGD-like optimizers having some properties similar to overfitting, in which the predicted values are overconfident, the pointwise early stopping algorithm for confidence scores selectively estimates the uncertainty of highly confident points in deep neural classifiers [15]. Additionally, the Monte Carlo dropout (MC-dropout) method casts dropout training in DNNs as approximate Bayesian inference and samples at the test phase, and then applies variance statistics for multiple dropout-enabled forward passes [16].

Most of the uncertainty estimation models mentioned above mainly consider the predictive uncertainty that comes from models and their training processes, known as "epistemic uncertainty" [16]. However, further predictive uncertainty derives from natural randomness such as noisy data and labels, class overlap, incomplete features, and other unknown factors; this is known as "aleatoric uncertainty" [17], which is inherent in the task and cannot be explained away with further data. Fortunately, a unified Bayesian deep learning framework has been proposed by [17], which can help us to capture an accurate understanding of aleatoric and epistemic uncertainty for per-pixel depth regression and semantic segmentation tasks. Forward passes in DNNs can be considered to be state transformations of a dynamical system, which can be defined by a neural-network-parameterized ordinary differential equation (ODE) [18]. An ODE is a deterministic expression, so it cannot obtain the epistemic uncertainty message.

Recently, a novel SDE-Net for uncertainty estimation of DNNs has been proposed to capture epistemic uncertainty using Brownian motion or the Wiener process [19,20], which are widely used to model uncertainty or randomness in mathematics, physics, economics, and other disciplines [21,22]. SDE-Net uses two separate neural networks (NNs): the drift net f is designed to control the system in order to achieve a good predictive accuracy for in-distribution (ID) data, while the diffusion net g is used to control the variance of the Brownian motion based on the ID or OOD regions. SDE-Net can not only explicitly model aleatoric uncertainty and epistemic uncertainty in its predictions for classification and regression tasks, but also does not need to specifically model prior distributions and infer posterior distributions as in BNNs. SDE-Net can achieve good performance and uncertainty estimation between ID and OOD data; however, in practice, ID data are generally encountered with noise or high missing rates. The purpose of this paper is to explore how to effectively deal with ID data with noise or high missing rates for SDE-Net. Our idea is to first fix the noisy ID data with a completion net, and then process the completed data with SDE-Net for uncertainty estimation. The components of SDE-Net are described in Figure 1a, and the resolution of SDE-Net's problems is explained in Figure 1b. Figure 1 shows that SDE-Net lacks consideration of the general situation in a wider field—that is, ID data with noise or high missing rates in practice.

To handle OOD data, existing studies either use confidence scores to determine whether samples are ID or OOD [23], or use a new confidence loss on a sharp predictive distribution for ID data and a flat predictive distribution for OOD data [24]. These new methods adopt loss functions to produce deterministic results. However, Dirichlet distribution—which allows high uncertainty for OOD data, but is only applicable to classification tasks—is parameterized over categorical distributions [25].



**Figure 1.** Illustration of SDE-Net and the problems it faces. (**a**) Components of SDE-Net. For ID data, SDE-Net is dominated by the drift net in order to achieve good predictive accuracy. For OOD data, SDE-Net is dominated by the diffusion net in order to generate high diffusion for characterizing model uncertainty. (**b**) Flowchart explaining how to resolve the problems faced by SDE-Net.

To handle noisy ID data, recent contributions indicate that the regularization technique dropout can degrade DNNs' training performance on noisy data without compromising generalization on real data [26]. More importantly, the dropout method has been proven to be a Bayesian approximation, and can represent model uncertainty in deep learning [27]. Moreover, the latest study establishes the first benchmark of controlled real-world label noise from the internet, and can conduct the largest study into understanding DNNs trained on noisy labels across different settings—such as noise levels, noise types, network architectures, and training settings. Thus, this method can be studied further for uncertainty estimation in deep learning. Of course, there are many other methods that deserve further study for uncertainty estimation, such as label cleaning/correction, example weighting, data augmentation, etc.

However, NP methods can combine the advantages of GPs in flexibility and neural networks with high precision; thus, we propose to add neural processes (NPs) [28] to SDE-Net in order to improve the its accuracy with noisy ID data, where NP variants include conditional neural processes (CNPs) [29], attentive neural processes (ANPs) [30], and convolutional conditional neural processes (ConvCNPs) [31]. We use the abbreviation vNPs to represent the NP family, which includes vanilla NPs and NP variants. The combination of vNPs and SDE-Net (vNPs–SDE) is motivated by the permutation invariance or equivariance properties of vNPs for ID data with noise or high missing rates in SDE-Net.

CNPs define distributions over functions given a set of observations, and the dependence of a CNP on the observations is parameterized by a neural network, which is invariant under permutations of its inputs. Meanwhile, vanilla NPs are a generalization to other NP variants, and the vanilla NPs generate fixed-length latent variables via a permutation invariant function. Moreover, ANPs apply an attention mechanism in order to compute the weights of each key with respect to the query [32]. ConvCNPs can model translation equivariance in the data and embed datasets into an infinite-dimensional function space. Although convolutional neural networks (CNNs) can also apply translation equivariance to time series or image tasks [33,34], the translation equivariance of CNNs models is not straightforward to generalize to the NP family in the same way. This is because CNNs models need image pixels in order to form a regularly spaced grid, while NPs perform on partially observed context sets in order to embed them into a finite-dimensional vector space.

To summarize, there are two main contributions of this paper:

- Considering the translation equivariance properties of ConvCNPs, the implementation
  of the vNPs–SDE model with ConvCNPs can effectively handle ID data with missing
  rates for 1D regression and 2D image classification tasks.
- Applying the property of permutation invariance, the implemented vNPs–SDE model with CNPs or ANPs surpasses BBP, MC-dropout, and vanilla SDE-Net in multidimensional regression tasks with high missing rates by most metrics.

The rest of this paper is organized as follows: Section 2 describes materials and methods for uncertainty estimation in deep learning models with noisy ID data. Section 3 presents the implementation of the proposed vNPs–SDE model for different tasks in deep learning. Section 4 demonstrates the results of ConvCNPs–SDE model for 1D regression and 2D image classification tasks with high missing rates, and of the CNPs–SDE and ANPs–SDE models for multidimensional regression tasks with high missing rates. Section 5 presents the discussion of the experimental results, and Section 6 presents the conclusions and implications for future work.

## 2. Materials

In this section, we mainly introduce the concepts to be used in this paper, such as stochastic processes and NPs and the relationship between them.

### 2.1. Definition of the Neural Processes Family

Neural processes as stochastic processes. For each finite sequence  $x_{1:n} = (x_1, \dots, x_n)$  with  $x_i \in X$ , the finite-dimensional marginal joint distribution over the function f values can be defined as  $Y_{1:n}(f(x_1), \dots, f(x_n))$ . For example, in the popular Gaussian processes (GPs) model, the joint distributions are multivariate Gaussian distributions parameterized by a mean and a covariance function.

As stated by the Kolmogorov extension theorem [20], two necessary conditions— (finite) exchangeability, and consistency for marginal joint distributions  $\rho_{x_{1:n}}$  of  $(f(x_1), \dots, f(x_n))$ —can be sufficient to define a stochastic process.

**Property 1** (Exchangeability) [28]. *If for each finite n element,*  $x_{1:n}$ *,*  $\pi$  *represents a permutation of*  $(1, \dots, n)$ *, then:* 

$$\rho_{x_{1:n}}(y_{1:n})\rho_{x_{1},\cdots,x_{n}}(y_{1},\cdots,y_{n}) = \rho_{x_{\pi(1)},\cdots,x_{\pi(n)}}(y_{\pi(1)},\cdots,y_{\pi(n)}) =: \rho_{\pi(x_{1:n})}(\pi(y_{1:n}))$$
(1)

**Property 2** (Consistency) [28]. If  $1 \le m \le n$ , and we marginalize out a part  $(y_{m+1}, \dots, y_n)$  of the sequence  $(y_1, \dots, y_n)$  of Y, the resulting marginal distribution is the same as that defined in the original sequence. That is:

$$\rho_{x_{1:m}}(y_{1:m}) = \int \rho_{x_{1:n}}(y_{1:n}) dy_{m+1:n}$$
(2)

Exchangeability and consistency can define a stochastic process, assuming a stochastic process f can be parameterized by a global and high-dimensional random vector z, so we can define a generative model:

$$p(z, y_{1:n} | x_{1:n}) = p(z) \prod_{i=1}^{n=1} N(y_i | g(x_i, z), \sigma^2)$$
(3)

NPs contain a latent variable z to capture stochastic process f and global uncertainty. An NP model is composed of three key components:

- (1) Encoder: The encoder *E* of NPs has two paths—a deterministic path and a latent path. In the deterministic path, each context pair  $(x, y)_i$  is passed through a multi-layer perceptron  $(MLP_{\theta})$  to produce a deterministic representation  $r_i$ . In the latent path, a latent representation  $s_i$  is generated by passing through each context pair  $(x, y)_i$  to another  $MLP_{\psi}$ . Thus, the purpose of encoder *E* is to convert the input space into deterministic or latent representation space, where the input space represents *n* context points  $C = \{(x, y)_i\}_{i=1}^n$ , and the representation space produces  $r_i = MLP_{\theta}((x, y)_i)$  and  $s_i = MLP_{\psi}((x, y)_i)$  for each of the pairs  $(x, y)_i$ .
- (2) Aggregator: Aggregator *a* aims to summarise the *n* global representations  $r_{1...n}$  and  $s_{1...n}$ . The simplest operation of aggregator *a* is the mean function  $m = a(m_i) = \frac{1}{n} \sum_{i=1}^{n} m_i$ , which can ensure order invariance and perform well in practice. For the deterministic path, *a* is applied to  $r_{1...n}$  to produce the deterministic code  $r_C$ . For the latent path, however, we are interested in achieving an order-invariant global latent representation, so we apply *a* to  $s_{1...n}$  to produce the latent code  $s_C$ , which can parameterize the normal distribution  $z \sim N(\mu(s_C), I\sigma(s_C))$  for the latent path.
- (3) Decoder: In decoder *D*, the sampled global latent variables *z* and  $r_C$  are concatenated alongside the new target locations  $x_T$  as inputs, and finally passed through *D* to produce the predictions  $\hat{y}_T = D(x_T, r_C, z)$  for the corresponding values of  $f(x_T) = y_T$ . We parameterize decoder *D* as a neural network.

NPs are a generalization of ANPs and CNPs, but CNPs lack a latent variable *z* that allows for global sampling.

The architectures of CNPs and ANPs are described in Figure 2.



**Figure 2.** Architectures of CNPs and ANPs. (a) Components of a CNP model: the encoder of CNPs is composed of a deterministic path to generate a representation r; the decoder of CNPs uses the r and target  $x_T$  to produce the mean and the Std. (b) Components of an ANP model: the encoder of ANPs is composed of a deterministic path to generate a representation  $r^*$  and a latent path to generate latent variable z; the decoder of ANPs uses the  $r^*$ , z, and target  $x_T$  to produce the mean and the Std.

Considering the CNPs model in Figure 2a, compared with the NPs and ANPs models, CNPs lack a latent path, and only produce the deterministic representation  $r_i = \text{MLP}_{\theta}((x, y)_i)$ for each of the context pairs  $(x, y)_i$  in the encoder. Then, the aggregator *a* of the CNPs summarises the *n* global representations  $r_{1...n}$  to produce the deterministic code  $r_C$ . Finally,  $r_{C}$  is concatenated alongside the new target locations  $x_{T}$  as an input, and passed through the neural network MLP<sub> $\varphi$ </sub> to produce the predictions  $\hat{y}_T = \text{MLP}_{\varphi}(x_T, r_C)$  in the decoder.

Considering the ANPs model in Figure 2b, compared with the NPs model, ANPs add an attention mechanism to increase the accuracy of the NPs. Assume that there are *n* key–value pairs  $(K, V) = (x_i, y_i)_{i=1}^n$ , where  $K \in \mathbb{R}^{n \times d_k}$ ,  $V \in \mathbb{R}^{n \times d_v}$ , and *m* query  $x_T = Q \in \mathbb{R}^{m \times d_k}$ . There are several attention mechanisms, such as uniform, laplace, dot product, and multihead:

- Uniform(Q, K, V) $\frac{1}{n}\sum_{i=1}^{n} x_i;$
- Laplace $(Q, K, V)WV \in \mathbb{R}^{n \times d_v}, W_i.softmax\left(\left(- \| Q_i K_j \|_1\right)_{j=1}^n\right) \in \mathbb{R}^n;$ DotProduct $(Q, K, V)softmax\left(\frac{QK^T}{\sqrt{d_k}}\right)V \in \mathbb{R}^{m \times d_v};$ MultiHead(Q, K, V)concat (head<sub>1</sub>, · · · , head<sub>H</sub>) $W \in \mathbb{R}^{m \times d_v},$

where head<sub>h</sub>DotProduct  $\left(QW_{h}^{Q}, KW_{h}^{K}, VW_{h}^{V}\right) \in \mathbb{R}^{m \times d_{v}}$ .

Compared with NPs, the advantage of an ANPs model is to incorporate attention mechanisms into the NPs model. In short, self-attention of the encoder in the deterministic and latent paths can be implemented via a multilayer perceptron (MLP), which is applied to the context points to get the representations  $r_{1...n}$ , and then target input  $x_T$  attends to  $r_{1...n}$  and  $x_{1...n}$  with cross-attention to predict the target output  $r^*$ . In the aggregator and the decoder, ANPs and NPs have similar operations.

First of all, for the definition of ConvCNPs, the translation equivariance is defined in Property 3.

**Property 3** (Translation equivariance) [31]. Assume H is a function space on X, and T and T' can be defined:

$$T: X \times \mathcal{Z} \to \mathcal{Z}, T_{\tau} Z = ((x_1 + \tau, y_1), \cdots, (x_m + \tau, y_m))$$
$$T': X \times H \to H, T'_{\tau} h(x) = h(x - \tau)$$

The mapping  $\Phi: \mathcal{Z} \to H$  is called translation equivariance, if  $\Phi(T_{\tau} Z) = T'_{\tau} \Phi(Z)$  for all  $\tau \in X$  and  $Z \in \mathcal{Z}$ .

**Theorem 1.** Assume a collection  $\mathcal{Z}'_{\leq M} \subseteq \mathcal{Z}_{\leq M}$ , which has multiplicity K. If the function  $\Phi: \mathcal{Z}'_{\leq M} \to C_b(X, Y)$  is permutation invariant, translation equivariant, and continuous, then  $\Phi$  has a representation as follows [31]:

$$\Phi(Z) = \rho(E(Z)), \ E((x_1, y_1), \cdots, (x_m, y_m)) = \sum_{i=1}^m \Phi(y_i)\psi(-x_i)$$

For continuous and translation-equivariant  $\rho$  :  $H \rightarrow C_b(X, Y)$ , continuous  $\Phi: Y \to \mathbb{R}^{K+1}$ , and  $\psi: X \to \mathbb{R}$ , where *H* is a function space, the function  $\Phi$  is called ConvDeepSet.

The key considerations of  $\phi$ ,  $\psi$ , and  $\rho$  for  $\Phi$  of ConvCNPs are:

- (1) Setting  $\psi$  to be a positive definite reproducing kernel Hilbert space (RKHS) [35].
- (2) Setting  $\phi(y) = (y^0, y^1, \cdots, y^K)$  [36].
- (3) Setting  $\rho$  to be a CNN.

The ConvCNPs can be represented by a conditional distribution, as follows:

$$p(Y \mid X, Z) = \prod_{n=1}^{N} p(y_n \mid \Phi_{\theta}(Z)(x_n)) = \prod_{n=1}^{N} \mathcal{N}(y_n; \mu_n, \Sigma_n)$$
(4)

where  $(\mu_n, \Sigma_n) = \Phi_{\theta}(Z)(x_n)$ .

For an on-the-grid version of  $\rho(\cdot)$ , a CNN is firstly applied to E(Z), and then an MLP can map the output at each pixel in the target set to  $\mathbb{R}^{2C}$ . To summarize, the on-the-gird algorithm is given by:

$$\left(\mu,\sigma^{2}\right) = \underbrace{\operatorname{CNN}}_{\rho} \left( \underbrace{\operatorname{Conv}(M_{c})}_{\text{density channel}} ; \operatorname{Conv}(M_{c} \odot I) / \underbrace{\operatorname{Conv}(M_{c})}_{\text{multiplies by }\psi \text{ and sums}} \right]^{T} \right)$$
(5)

where  $(\mu, \sigma^2)$  are means and variances,  $\rho$  is implemented with a CNN, and *E* is produced by the mask  $M_c$  and a convolution operation.

#### 2.2. Definition of SDE-Net

Neural ordinary differential equation (ODE-Net): Neural nets such as residual networks (ResNet) [37], normalizing flows [38], and recurrent neural network decoders [39] map an input x to an output y through a sequence of hidden layers; the hidden representations can be viewed as the states of a dynamical system:

$$x_{t+1} = x_t + f(x_t, t)$$
 (6)

where  $t \in \{0 \cdots T\}$  is the index of the layer, and  $x_t \in \mathbb{R}^D$  is the hidden state at neural network layer *t*. The equation can be reorganized as  $\frac{x_t + \Delta t - x_t}{\Delta t} = f(x_t, t)$ , where  $\Delta t = 1$ . If we assume that  $\Delta t \to 0$ , then we can obtain the parameterized continuous dynamics of the hidden units, which apply an ODE specified by a neural network:

$$\lim_{\Delta t \to 0} \frac{x_t + \Delta t - x_t}{\Delta t} = \frac{\mathrm{d}x_t}{\mathrm{d}t} = f(x_t, t, \theta)$$
(7)

The solution of an ODE can be computed using a black-box differential equation solver to evaluate the hidden unit state wherever necessary. However, ODE-Net is a deterministic model for predictions, and cannot model epistemic uncertainty. To overcome this disadvantage, the novel SDE-Net model is proposed to characterize a stochastic dynamical system for capturing epistemic uncertainty with Brownian motion, which is widely used to model the randomness of the motion of atoms or molecules in physics.

SDE-Net: A standard Brownian motion term is added to Equation (7) to form a neural SDE dynamical system. The continuous-time dynamical system is expressed as follows:

$$dx_t = f(x_t, t)dt + g(x_t, t)dW_t$$
(8)

where  $g(x_t, t)$  indicates the variance of the Brownian motion, and represents the epistemic uncertainty of the dynamical system. However, a standard Brownian motion  $W_t$  is a stochastic process, which follows the three properties: (a)  $W_0 = 0$ ; (b)  $\nabla W = W_t - W_s$  is N(0, t - s) for all  $t \ge s \ge 0$ ; and (c) for any two different time intervals, the increments  $\nabla W_1$  and  $\nabla W_2$  are independent random variables.

More importantly,  $f(x_t, t)$  and  $g(x_t, t)$  in Equation (8) can be represented by NNs to construct SDE-Net. Where  $f(x_t, t)$  is used as the drift net to control the system in order to achieve good predictive accuracy and aleatoric uncertainty, and  $g(x_t, t)$  is utilized as the diffusion net to represent the epistemic uncertainty of the dynamical system.  $f(x, t; \theta_f)$  and  $g(x_0; \theta_g)$  must both be uniformly Lipschitz continuous. This can be satisfied by

using Lipschitz nonlinear activations in the network architectures, such as ReLU, sigmoid, and Tanh.

## 3. Proposed Methods

Section 3.1 shows the architecture of vNPs–SDE net, which is implemented by different NPs for specific deep leraning tasks. Section 3.2 presents the objective function of vNPs–SDE net for uncertainty estimates. The implementation algorithms of the ConvCNPs–SDE-, ANPs–SDE-, and CNPs–SDE-Net are described in Section 3.3.

#### 3.1. The Architecture of vNPs–SDE-Net

In Section 3.1.1, for synthetic 1D regression and 2D image classification tasks, we firstly apply ConvCNPs to complete the ID data with high missing rates, and then use SDE-Net to quantify the uncertainty of the noisy ID dataset.

In Section 3.1.2, for multidimensional regression tasks, we replace the downsampling NNs in SDE-Net with the encoder of CNPs or ANPs in order to encode the regression data as latent representation r, then use the drift net and diffusion net of SDE to deal with r, and finally apply the decoder of CNPs or ANPs to substitute the fully-collected NNs in SDE-Net.

## 3.1.1. vNPs-SDE-Net for Synthetic 1D Regression and 2D Image Classification Tasks

For synthetic 1D regression and 2D image classification tasks, the latest research [31] has extensively compared the ConvCNPs model to the CNPs and ANPs models, and proved that the ConvCNPs model with translation equivariance can improve performance in off-the-grid synthetic 1D datasets or on-the-grid image datasets. The reason for the lack of comparision between vanilla NPs models and the ConvCNPs model is that the ANPs model with attention mechanisms is usually superior to vanilla NPs, which suffer a fundamental disadvantage of underfitting and give inaccurate prediction values at the observed context points on which they condition [30], so the CNPs and ANPs models can be viewed as preeminent representatives of the NPs family.

In Figure 3, we introduce the architecture of the ConvCNPs–SDE model, which contains ConvCNPs and the SDE-Net model for training synthetic 1D regression and 2D image classification tasks.



Figure 3. Architecture of the ConvCNPs-SDE model. ID datasets were used to train ConvCNPs-Net and SDE-Net.

For the ConvCNPs-Net model, we apply ID datasets to train ConvCNPs and SDE-Net. Where ConvCNPs select all observed context points as signal channel  $SM_c \odot I$ , suppose I stands for the image and  $M_c$  denotes density channel  $DM_c$ .

We can concatenate S and D to form [S, D], which means that "there is a point at this position". Then a convolutional neural network (CNN) is applied to a normalized [S, D] to produce Means and Vars, which can be used to generate a continuous multivariate normal distribution.

For the SDE-Net model used in 1D regression tasks, we adopt upsampling with a linear layer. The drift net uses a fully connected linear layer, and the linear layer adopts ReLU activation without batch normalization. The diffusion net can apply multiple linear layers and ReLU activations, and it has one output and employs sigmoid activation for the last layer.

For the SDE-Net model used in 2D image classification tasks, we use a downsampling layer with multiple convolutional layers for extracting features; each layer uses ReLU activation and batch normalization. The drift net consists of convolutional layers, and the input channel of each layer sets aside an extra position for layer depth, which signifies the number of discretization points of the stochastic process. The diffusion net is the same as the drift net, except for the last layer, which uses sigmoid activation function to return 0 or 1 for the diffusion net, while the drift net outputs based on the number of output channels [19].

The above mainly introduces the model structure of ConvCNPs and SDE-Net in the training phase for synthetic 1D regression and 2D classification tasks with ID datasets.

Figure 4 shows that the ConvCNPs model can be utilized to complete masked ID datasets, and the recovered ID datasets can be more effectively and accurately recongnized by SDE-Net.



**Figure 4.** The process architecture of the ConvCNPs–SDE model for masked ID datasets. ConvCNPs-Net is used to complete the masked ID data, and then the completed data is processed with SDE-Net to produce Means and Vars.

## 3.1.2. vNPs–SDE-Net for Multidimensional Regression Tasks

For 1D regression and 2D classification tasks, the ConvCNPs model has advantages over other members of the NPs family. For multidimensional regression tasks, however, it is difficult to apply the property of translation equivariance, due to its uncertain number of dimensions, so we try to adopt ANPs and CNPs models.

For example, in the encoder of the CNPs model [29], observed context points  $(x, y)_1, \ldots, (x, y)_i$  are passed through an MLP to generate representations  $r_1, \ldots, r_i$ , and then mean function *m* is employed to  $r_1, \ldots, r_i$  to produce the global deterministic representation  $r_c$ . Specifically, the encoder in CNPs has six latent fully connected linear layers with ReLU activation, and the number of neurons in the six latent layers is equal to the dimension *n* of the regression dataset. For the decoder of the CNPs model, we still apply an MLP with five fully connected linear layers and ReLU activation; the generated representation  $r_c$  is concatenated with the target data  $x_T$  and together passed through the defined decoder MLP to produce the parameters Means and Vars for multivariate normal distribution.

For the encoder of the ANPs model [30], we still adopt an MLP with six fully connected linear layers to replace the self-attention mechanism for the deterministic path, and apply

three fully connected linear layers for the latent path. The mean aggregation is replaced by a multihead cross-attention mechanism, and the number of heads is 10. Thus, in the deterministic path, each context point  $(x, y)_i$  is passed through an MLP to produce representation  $r_i$ ; the target query  $x_T$  attends to the *i* key–value pairs  $(x_i, r_i)$ , and assigns weights  $w_i$  to each pair in order to generate a representation  $r^* = \sum w_i r_i$ .

For the latent path of the ANPs model, a representation  $r_C$  is generated in a similar manner to the encoder of the CNPs model. Global latent  $r_C$  can be utilized to parameterise a multivariate normal distribution, which can model different realisations of the data-generating stochastic process, and sample from the distribution to produce the latent variable *z*, corresponding to one realization of the stochastic process.

For the decoder of the ANPs model,  $r^*$  and z are concatenated with  $x_T$  and passed through an MLP with six fully connected linear layers to produce the parameters Means and Vars of multivariate normal distribution.

In Figure 5a, the ID dataset is used to train the CNPs–SDE or ANPs–SDE models; the encoder and decoder are from the CNPs and ANPs models, respectively, so as to generate the parameters Means and Vars of normal distribution. In Figure 5b, a masked ID dataset is utilized to test the performance of the constructed vNPs–SDE model when facing the ID dataset with noise or a high missing rate.





The ANPs model with attention mechanisms is more expressive and accurate than the NPs and CNPs models. At test time, the computational time complexity of the NPs and CNPs models is O(n + m), because each of n context points passes through the MLP of the encoder to generate  $r_1, \ldots, r_n$  for producing  $r_c$ , and then  $r_c$  is incorporated with each of m target points in the decoder to generate m predicted values. However, the computational time complexity of the ANPs model increases from O(n + m) to O(n(n + m)), since the self-attention is applied to n contexts, and m target points are used to compute weights for all of n contexts.

# 3.2. The Objective Function of the vNPs–SDE-Net for Uncertainty Estimates

The objective function for training the vNPs-SDE-Net model is:

$$\min_{\theta_{vNP}} E_{x_0 \sim P_{train}} E(LogP(x_0)) + \min_{\theta_f} E_{x_0 \sim P_{train}} E(L(x_T)) 
+ \min_{\theta_g} E_{x_0 \sim P_{train}} g(x_0; \theta_g) + \min_{\theta_g} E_{\widetilde{x}_0 \sim P_{OOD}} g(\widetilde{x}_0; \theta_g) 
s.t. dx_t = \underbrace{f(x_t, t; \theta_f)}_{drift \ neural \ net} dt + \underbrace{g(x_0; \theta_g)}_{diffusion \ neural \ net} dW_t$$
(9)

where  $LogP(\cdot)$  in Equation (9) is the log-likelihood loss function as a reconstruction term for the ConvCNPs and CNPs models. However, for the ANPs model, the first item of the objective function in Equation (9) is the evidence lower bound (ELBO), which includes a reconstruction term and a Kullback–Leibler divergence (KL) term. This is because the ANPs model has a latent path through which to generate latent variable *z* for modelling uncertainty.  $L(\cdot)$  is the loss function dependent on the task, such as cross-entropy loss for classification tasks and log-likelihood loss for regression tasks; *T* is the terminal time of the stochastic processes;  $P_{train}$  is the distribution of the training data; and  $P_{OOD}$  represents the OOD data. The OOD data can be obtained by adding additive Gaussian noise to the noisy inputs  $\tilde{x}_0 = x_0 + \epsilon$ , and then distributing the inputs according to the convolved distribution.

Once an vNPs–SDE-Net has been trained, we can obtain multiple random realizations of the SDE-Net in order to get samples  $\{x_T\}_{m=1}^M$ , and then compute the two uncertainties from them. The aleatoric uncertainty is given by the expected predictive entropy  $\mathbb{E}_{p(x_T|x_0, \theta_{f,g})}[\mathcal{H}[p(y|x_T)]]$  in classification, and by the expected predictive variance  $\mathbb{E}_{p(x_T|x_0, \theta_{f,g})}[\sigma(x_T)]$  in regression. The epistemic uncertainty is given by the variance of the final solution  $\operatorname{Var}(x_T)$ .

After the vNPs and SDE-Net models are trained, suppose that the ID dataset *I* with missing rate MR and mask = Bernoulli (1-MR), so the masked ID dataset *I* can be expressed as mask \* I. The test performance of vNPs–SDE for the masked ID dataset can be processed as follows:

- (1) Completed\_I = vNPs (mask \* I)
- (2) Means, Vars = SDE-Net(Completed\_*I*)

Since our purpose is to perform supervised learning and uncertainty quantification, the simple Euler–Maruyama method with fixed step size is adopted for model training. Hence, the time interval [0, *T*] is divided into *N* subintervals, and SDE can be simulated as:

$$x_{k+1} = x_k + f\left(x_k, \ t; \theta_f\right) \Delta t + g\left(x_0; \theta_g\right) \sqrt{\Delta t} \ Z_k \tag{10}$$

where  $Z_k \sim N(0,1)$  and  $\Delta t = T/N$ . The number of steps for solving the SDE can be regarded equivalently as the number of layers in the NNs. Moreover, the training of SDE-Net is actually the same as in NNs. The vNPs–SDE model is optimized in the Algorithms in the following sections.

### 3.3. The Implementation of vNPs-SDE-Net

In this section, vNPs are implemented with ConvCNPs for 1D regression and 2D image classification tasks in Algorithm 1, and vNPs are realized with CNPs and ANPs for multidimensional regression tasks in Algorithm 2.

#### 3.3.1. The Implementation of vNPs-SDE-Net with ConvCNPs

Assume an ID dataset  $\{(x, y)_i\}_{i=1}^n \sim P(x, y)$  for 1D regression or 2D image classification tasks, and then sample a minibatch of *m* data, which includes inputs  $X^m$  and targets  $Y^m$  from the ID dataset:  $(X^m, Y^m) \sim p_{ID}(x, y)$ .

For training the ConvCNPs model, suppose that the context rate (*CR*) is 80%, so the context dataset  $(X_c^m, Y_c^m)$  is composed of 80% of *m* datasets, and the sampled *m* data are viewed as the target dataset  $(X_t^m, Y_t^m)$ . The output of the ConvCNPs model is a multivariate normal distribution, so the predicted Means and Vars can be called by the ConvCNPs model, the purpose of which is to complete the masked ID dataset.

For the SDE-Net model, the purpose of the training of the drift net is to fit the ID dataset. Meanwhile, for the training diffusion net *g*, the dataset from ID or OOD is respectively marked with labels 0 and 1, and the purpose of *g* is to distinguish whether the dataset comes from ID or OOD, so the purpose of training *g* is to minimize or maximize the binary cross-entropy loss function for the ID dataset or the OOD dataset, respectively.

For 1D regression tasks, the specific settings of SDE-Net include a layer depth of four; the upsampling net has a linear layer with a [1–50] architecture, the drift net has a fully connected linear layer with a [50–50] architecture, and the diffusion net has a [50–100–100–1] fully connected linear layer. ReLU is the activation function. The number of training epochs is 1000 and the optimizer is an SGD. The learning rate for the diffusion

net is 0.01, while for the drift net it is 1e-4, and the learning rate is multiplied by 0.1 when the number of epochs reaches 60 and the momentum and weight decay are 0.9 and 5e-4, respectively. We reshape the generated ID dataset to (batch\_size, total \_points), where batch\_size = 15 and total\_points = 100. Assume the MR is given, so the remaining training dataset is (1-MR) × total\_points for the 1D tasks.

Algorithm 1 Implementation of the ConvCNPs-SDE model

**Inputs:** ID dataset  $p_{ID}(x, y)$ ; *CR* and MR are the context rate and missing rate, respectively; *ccnps* represents the ConvCNPs model of vNPs for completing the ID dataset;  $h_1$  is the downsampling net for 2D image classification tasks or the upsampling net for 1D regression tasks;  $h_2$  is the fully connected net; *f* represents the drift net and *g* represents the diffusion net; *t* is the layer depth;  $L_1$  is the cross-entropy loss function,  $L_2$  is the log-likelihood loss function, and  $L_3$  is the binary cross-entropy loss function.

Outputs: Means and Vars

for #training iterations do

- 1. Sample a minibatch of *m* data:  $(X^m, Y^m) \sim p_{ID}(x, y)$ ;
- 2. **if** for 1D regression task:

3. Context points  $(X_c^m, Y_c^m)$  are generated from sampled target points  $(X_t^m, Y_t^m)$  based on *CR*, where  $(X_t^m, Y_t^m)$  equals  $(X^m, Y^m)$ ;

- 4. Forward through the ConvCNPs model:  $Y_{dist} = ccnps(X_c^m, Y_c^m, X_t^m)$ ;
- 5. Forward through the upsampling net of the SDE-Net block:  $X_0^m = h_1(X^m, Y^m)$ ;
- 6. **else** for 2D image classification task:
- 7. Forward through the ConvCNPs model:  $Y_{dist} = ccnps(X^m, Y^m)$ ;
- 8. Forward through the downsampling net of the SDE-Net block:  $X_0^m = h_1(X^m, Y^m)$ ;
- 9. **for** k = 0 to t 1 **do**
- 10. Sample  $Z_k^m \sim \mathcal{N}(0,1)$ ;

 $X_{k+1}^{m} = X_k^{m} + f(X_k^{m}, t)\Delta t + g(X_0^{m})\sqrt{\Delta t}Z_k;$ 

- 11. end for
- 12. Forward through the fully connected layer of the SDE-Net block:  $Y_f^m = h_2(X_k^m)$ ;
- 13. Update  $h_1$ ,  $h_2$  and f by  $\nabla_{h_1, h_2}$  and  $f \frac{1}{m} L_1(Y_f^m, Y^m)$ ;
- 14. Update *ccnps* by  $\nabla_{ccnps} \frac{1}{m} L_2(Y_dist.means, Y_t^m)$ ;
- 15. Sample a minibatch of *m* data from ID:  $(X^m, 0) \sim p_{ID}(x, y)$ ;
- 16. Sample a minibatch of *m* data from OOD:  $(\tilde{X}^m, 1) \sim p_{OOD}(x, y)$ ;
- 17. Forward through the downsampling or upsampling nets of the SDE-Net block:
- $X_0^m, \tilde{X}_0^m = h_1(X^m), h_1(\tilde{X}^m);$
- 18. Update g by  $\nabla_g \frac{1}{m} L_3(g(X_0^m), 0) \nabla_g L_3(g(\widetilde{X}_0^m), 1);$
- for #testing iterations do
- 19. Evaluate the of ConvCNPs–SDE model;
- 20. Sample a minibatch of *m* data from ID:  $(X^m, Y^m) \sim p_{ID}(x, y)$ ;
- 21. *mask* = Bernoulli (1-MR)
- 22.  $masked_X^m = mask * X^m$ ;
- 23.  $completed_X^m = ccnps(masked_X^m);$
- 24. Means, Vars = SDE-Net(*completed*\_ $X^m$ );

For the 1D regression tasks, the specific settings of the ConvCNPs–SDE model include the ConvCNPs net having four 1D convolution layers, which can be described as  $[in\_channel, out\_channel, kernel\_size$ , and *stride*, *padding*], and specifically contain {[3, 16, 5, 1, 2], [16, 32, 5, 1, 2], [32, 16, 5, 1, 2], and [16, 2, 5, 1, 2]}. The density is 25 and K = 1 for  $\phi(y) = (y^0, y^1, \dots, y^K)$ , and the RBF is chosen for covariance  $\psi(y)$ ; thus, these settings can satisfiy Theorem 1. The model settings of SDE-Net are the same as those defined in the previous paragraph, except for the upsampling layer, which concatenates x and y as inputs and has fully connected linear NNs with a [2–50] architecture. ReLU is the activation function. For training ConvCNPs–Net, the number of training epochs is 1000, Adam is the optimizer, and the learning rate and weight decay are 1e-3 and 1e-5, respectively. The setting of the training parameters is the same as for SDE-Net in the previous paragraph.

For the 2D image classification tasks, vanilla SDE-Net follows the settings of [19]. For the convolutional layer, the downsampling layers contain three 2D convolution (Conv2d) layers, which can be described as {[1, 64, 3, 1, 0], [64, 64, 4, 2, 1], and [64, 64, 4, 2, 1]} for the MNIST dataset and {[3, 64, 3, 1, 0], [64, 64, 4, 2, 1], and [64, 64, 4, 2, 1]} for the CIFAR10 dataset. The drift net contains two Conv2d layers with {[65, 64, 3, 1, 1], and [65, 64, 3, 1, 1]}, and the diffusion net has the same convolution layers as the drift net, but the diffusion net owns an extra linear connection [64–1] in the last layer. The fully connected layer of SDE-Net is [64–10]. In order to train the SDE-Net, the layer depth is 6 and the number of training epochs is 40, An SGD is used as the optimizer, the learning rates of diffusion net are 0.01 and 0.005 for MNIST and CIAFAR10, respectively, and the learning rates of the other nets are 0.1. The momentum and weight decay are 0.9 and 5e-4, respectively.

For the 2D image classification tasks, the ConvCNPs-Net of the ConvCNPs–SDE model selects all observed context points as signal channel  $SM_c \odot$  *Inputs*, assuming *Inputs* stands for the image and  $M_c$  denotes density channel  $DM_c$ , where  $M_c = Bernoulli(\frac{number \ of \ context \ pixel \ points}{number \ of \ total \ pixel \ points})$ 

and the number of context points is uniformly sampled from  $\left[\frac{number of total pixel points}{100}\right]$ 

<u>number of total pixel points</u>]. *S* and *D* are firstly processed by a Conv2d layer  $\{[1, 64, 9, 1, 4]\}$  for the MNIST dataset and  $\{[3, 64, 9, 1, 4]\}$  for the CIFAR10 dataset in order to generate *S'* and *D'*, and then we can concatenate them to form [S', D'], which is passed through a CNN, and finally the output of the CNN is transformed to a continuous function space for translation equivariance. The CNN firstly has a Conv2d layer with  $\{[128, 64, 1, 1, 0]\}$ , and then has eight Conv2d residual blocks—each residual block having two Conv2d layers with  $\{[64, 64, 5, 1, 2]\}$ —and finally owns the last Conv2d layer with  $\{[64, 2, 1, 1, 0]\}$  for the MNIST dataset and  $\{[64, 6, 1, 1, 0]\}$  for the CIFAR10 dataset. The training parameters include 20 training epochs; Adam is chosen as the optimizer, the batch size is 16, and the learning rate is 5e-4. The settings of SDE-Net in the ConvCNPs–SDE model are the same as in the previous paragraph.

### 3.3.2. The Implementation of vNPs-SDE-Net with CNPs or ANPs

For the multidimensional regression tasks, we apply vanilla CNPs or ANPs to represent vNPs. As described in Figure 5, the downsampling net and the fully connected layer are replaced by the encoder and decoder of the CNPs or ANPs models. The training and testing processes are described in Algorithm 2:

As the dimension of YearPredictionMSD is 90, we apply a fully connected DNN with a [91–90–90–90–90–90] architecture for the encoder of the CNPs model, and the decoder architecture of the CNPs model is [180–90–90–90–2]. For SDE-Net, the drift net has a fully connected linear layer with a [180–180] architecture, while the diffusion net has a [180–100–1] DNN architecture, and the layer depth is four. ReLU is the activation function. The number of training epochs is 60 and the optimizer is an SGD; the learning rate for the diffusion net is 0.01, while for the drift net it is 1e-4, and the learning rate is multiplied by 0.1 when the number of epochs reaches 30 and the momentum and weight decay are 0.9 and 5e-4, respectively.

For the ANPs of the ANPs–SDE model, the deterministic path and latent path of the encoder have fully connected DNN architecture—[91–90–90–90–90] and [91–90–90–180], respectively. The cross-attention encoder of (keys, queries) has linear layers with a [90–90] architecture, and a multihead attention mechanism is adopted in order to deal with the processed (keys, queries), where the embedded dimension is 90 and the number of parallel attention heads is 10. The decoder of the ANPs has a fully connected DNN with a [270–90–90–90–2] architecture. For vanilla SDE-Net, the architecture of the drift net is [270–270], while that of the diffusion net is [270–100–1], and the layer depth is four. ReLU is the activation function. The training parameter settings of the ANPs–SDE model are the same as those of the CNPs–SDE model.

Algorithm 2 Implementation of the CNPs-SDE or ANPs-SDE models

**Inputs**: ID dataset  $p_{ID}(x, y)$ ; MR is the missing rate of the ID dataset; the downsampling layer  $h_1$  is the encoder of the CNPs or ANPs models; f and g are the drift net and diffusion net, respectively;  $L_1$  is the negative log-likelihood loss function for the CNPs model or the ELBO for the ANPs model;  $L_2$  is the binary cross-entropy loss function; the fully collected layer  $h_2$  is the decoder of the CNPs or ANPs models to produce Means and Vars.

**Outputs**: Means and Vars **for** #training iterations **do** 

- 1. Sample a minibatch of *m* data:  $(X^m, Y^m) \sim p_{ID}(x, y);$
- 2. Forward through the downsampling net:  $d\_mean\_z = h_1(X^m)$  and  $X_0^m = (X^m, d\_mean\_z)$ ;
- 3. Forward through the SDE-Net block:
- 4. **for** k = 0 to n 1 **do**
- 5. Sample  $Z_k^{N_M} \sim N(0,1)$ ;
- 6.  $X_{k+1}^{m} = X_k^{m} + f(X_0^{m}, t)\Delta t + g(X_0^{m})\sqrt{\Delta t}Z_k;$
- 7. end for
- 8. Means, Vars =  $h_2(X_{k+1}^{m})$
- 9. Update  $h_1$ ,  $h_2$  and f by  $\nabla_{h_1, h_2}$  and  $f \frac{1}{m}L_1(\text{Means}, \Upsilon^m)$ ;
- 10. Sample a minibatch of *m* data from ID:  $(X^m, 0) \sim p_{ID}(x, y)$ ;
- 11. Sample a minibatch of  $(\tilde{X}^m, 1) \sim p_{OOD}(x, y)$ ;
- 12. Forward through the downsampling or upsampling nets of the SDE-Net block:
- $X_0^m, \tilde{X}_0^m = h_1(X^m), h_1(\tilde{X}^m);$

13. Update *g* by  $\nabla_g \frac{1}{m} L_2(g(X_0^m), 0) - \nabla_g L_2(g(\widetilde{X}_0^m), 1);$ 

- for #testing iterations do
- 14. Evaluate the CNPs–SDE or ANPs–SDE models;
- 15. Sample a minibatch of *m* data from ID:  $(X^m, Y^m) \sim p_{ID}(x, y)$ ;
- 16. *mask* = Bernoulli(1-MR);
- 17.  $masked_X^m = mask * X^m;$
- 18. Means, Vars = CNPs\_SDE ( $masked_X^m$ ) or ANPs\_SDE ( $masked_X^m$ ).

## 4. Results

Section 4.1 introduces the evaluation metrics. Section 4.2 presents the specific settings of the vNPs–SDE model and the experimental results.

#### 4.1. Evaluation Metrics

For an OOD detection task under both classification and regression settings, assume *P* represents real postive, *N* stands for real negative, *T* indicates true prediction, and *F* signifies false prediction. Thus, *TP* implies that predicted samples and real samples are postive, *FP* denotes that predicted samples are positive and real samples are negative, *FN* betokens that predicted samples are postive and real samples are positive, and *TN* symbolizes that predicted samples and real samples are negative.

We follow previous works such as [19,23], and apply several metrics [40] for the OOD detection task. Larger values of these metrics indicate better detection performance. *N* represents that the number of  $TPR \in [0.94, 0.96]$ .

$$TPR = \frac{TP}{TP + FN} \tag{11}$$

$$FPR = \frac{FP}{FP + TN} \tag{12}$$

$$TNR \text{ at } 95\% TPR = 1 - \frac{FPR}{N}$$
(13)

$$Precision = \frac{TP}{TP + FP}$$
(14)

$$Recall = \frac{TP}{TP + FN}$$
(15)

$$Accuracy = \frac{TP + TN}{TP + TN + FP + FN}$$
(16)

AUROC is used to denote the area under the receiver operating characteristic (*ROC*) curve; the horizontal axis of the *ROC* curve is represented by *FPR*, and the vertical axis is represented by *TPR*, so the points (*FPR*, *TPR*) on the *ROC* curve are coordinate points in the 2D Cartesian coordinate system. The probability of predicting the positive sample as postive is *p*1, and that of predicting the negative sample as positive is *p*2. Therefore, *AUROC* reflects the sorting ability of the classifier on the samples. In addition, *AUROC* is not sensitive to whether or not the sample categories are balanced, which is also the reason that *AUROC* is usually used to evaluate the classification performance of unbalanced samples.

$$AUROC = \frac{\sum_{i \in P} r_i - \frac{|P| \times (|P|+1)}{2}}{|P| \times |N|}$$
(17)

where *P* denotes a positive sample set, *N* signifies a negative sample set,  $|\cdot|$  stands for the number of samples, and  $r_i$  respresents the rank of element *i* in the total set (*P* + *N*), from the smallest to the largest, according to the predicted scores.

AUPR is used to represent the area under the precision-recall (*PR*) curve; the horizontal axis of the *PR* curve is represented by *Recall*, while the vertical axis is represented by *Precision*.

#### 4.2. vNPs–SDE Model for ID Dataset with MR

In this section we present the experimental results of the vNPs–SDE model for the ID dataset with MR. Specifically, the ConvCNPs-Net is performed for the synthetic 1D regression task in Section 4.2.1, the vNPs–SDE model is implemented with CNPs and ANPs for the multidimensional regression task in Section 4.2.2, and the ConvCNPs–SDE model is performed with the benchmark datasets MNIST and CIFAR10 in Sections 4.2.3 and 4.2.4, respectively.

We compare our vNPs–SDE model with the following methods for uncertainty estimates: (1) SDE-Net, which is the lastest non-Bayesian approach for uncertainty estimation in DNNs; (2) the approximate Bayesian method MC-dropout; and (3) the Bayesian approach BBP.

### 4.2.1. ConvCNPs–SDE-Net for Synthetic 1D Regression Tasks

In order to obtain the synthetic 1D dataset, the radial basis function (RBF) kernel was used for GPs to generate the synthetic 1D dataset [29,31]. The RBF kernel was  $k(x, x') = \sigma_1 * e^{-\frac{1}{2\sigma_2}(x-x')^2}$ , and we set the output scale to  $\sigma_1 = 0.5$  and the length scale to  $\sigma_2 = 0.5$ . We sampled 1500 *x*-axis data points from the *domain* [-2, 2] based on uniform distribution, and then the covariance of the GPs could be obtained from the sampled data points applied to the RBF, and the offset was 1e-5. If the means of the GPs are zeros and the GPs have noise—which has a mean of zero and standard deviation of 0.02—then we can generate 1500 *y*-axis GP data points for each batch. For the reproducibility of the experiment, the seed was 123.

In order to train SDE-Net, BBP, and ConvCNPs–SDE-Net, we firstly sorted the sampled *x*-axis data points from the smallest to largest, and then adjusted the generated *y*-axis data points corresponding to the *x*-axis to form the training GPs dataset  $\{(x, y)_i\}_{i=1}^{1500}$ .

The results are shown in Figure 6. The blue cross stands for training data, the black line represents the SDE-Net's posterior mean, the red line denotes the ConvCNPs–SDE-net's posterior mean, the yellow represents the BBP approach's posterior mean, and the shaded region signifies the mean  $\pm$  3 standard deviations. Since MC-Dropout produced poor results, it is not shown in Figure 6. Firstly, we find that as the MR increases, the ConvCNPs–SDE model can fit the remaining data points more accurately than vanilla SDE-Net and the BBP model. Secondly, the ConvCNPs–SDE model has smaller variance than the SDE-Net and BBP models. Lastly, and most importantly, the curve of the ConvCNPs–SDE model



is smoother than that of the vanilla SDE-Net and BBP models in almost all experiments, which is significant for those who have not sampled continuous *x*-axis samples in domain [-2, 2] to predict corresponding *y* values.

**Figure 6.** Visualizing the uncertainty of predictive distribution in regression tasks. (**a**–**f**) show that  $(1-MR) \times 100$  training data with MR = [0.0, 0.1, 0.3, 0.5, 0.7, 0.9], respectively, is used to train the SDE-Net, BBP, and the proposed ConvCNPs–SDE-Net.

#### 4.2.2. The CNPs-SDE and ANPs-SDE Models for Multidimensional Regression Tasks

We followed the methods of [19], applying the YearPredictionMSD multidimensional regression dataset for this experiment. The YearPredictionMSD dataset is used to predict the release year of a song from audio features [41], and the year ranges from 1922 to 2011. The dataset has 515,345 instances and 90 attributes. We followed the train/test split—the first 463,715 examples for training and the last 51,630 examples for testing.

We obtained randomized *mask* = Bernoulli (1-MR), where MR was chosen from [0.1, 0.3, 0.5, 0.7, 0.9] to compute the RMSE. The masked YearPredictionMSD dataset can be expressed as *mask* \* YearPredictionMSD. The BBP, MC-dropput, ANPs–SDE-Net, SDE-Net, and CNPs–SDE-Net models were run independently six times, and the results of the RMSE are shown in Table 1. Firstly, we can conclude that as the MR increases, the RMSE values gradually increase. Secondly, the CNPs–SDE model is more accurate than the BBP, MC-dropout, ANPs–SDE and SDE-Net models.

We applied the YearPredictionMSD dataset as ID data and the Boston Housing dataset as test OOD data. Table 2 shows the OOD detection performance for different models.

We report the average performance and standard deviation for five random initializations in Table 2. Because of the imbance of the test ID and OOD datasets, AUPR out is a better metric than AUPR in [19], so ANPs–SDE-Net is a more effective method than the other methods with respect to *AUPR in* and *AUPR out*, and the conclusions drawn from other metrics show that ANPs–SDE-Net is also comparable or superior to BBP, MC-dropout, and SDE-Net.

The experimental results of Tables 1 and 2 show that vNPs can not only help SDE-Net improve the accuracy of the ID dataset with MR, but also improve the OOD detection performance.

Model	Missing Rate	RMSE
BBP		$16.6 \pm 0.1$
MC-dropout		$12.4\pm0.4$
ANPs-SDE-Net	MR = 0.1	$9.3\pm0.6$
SDE-Net		$9.3\pm0.7$
CNPs-SDE-Net		$9.1\pm0.6$
BBP		$20.4\pm0.5$
MC-dropout		$15.2\pm0.6$
ANPs-SDE-Net	MR = 0.3	$13.1\pm0.9$
SDE-Net		$13.1 \pm 1.0$
CNPs-SDE Net		$12.9\pm0.8$
BBP		$23.5\pm0.4$
MC-dropout		$17.5 \pm 1.0$
ANPs-SDE-Net	MR = 0.5	$16.1 \pm 1.1$
SDE-Net		$16.1 \pm 1.2$
CNPs-SDE-Net		$15.8\pm1.0$
BBP		$26.3 \pm 1.1$
MC-dropout		$19.6 \pm 1.2$
ANPs-SDE-Net	MR = 0.7	$18.5 \pm 1.3$
SDE-Net		$18.5\pm1.4$
CNPs-SDE-Net		$18.2 \pm 1.2$
BBP		$28.8\pm1.3$
MC-dropout		$21.5\pm1.3$
ANPs-SDE-Net	MR = 0.9	$20.7 \pm 1.4$
SDE-Net		$20.7\pm1.6$
CNPs-SDE-Net		$20.4 \pm 1.3$

Table 1. ID dataset YearPredictionMSD with MR.

Table 2. OOD detection for regression on YearPredictionMSD + Boston Housing.

Model	#Parameters	RMSE	TNR at TPR 95%	AUROC	Detection Accuracy	AUPR In	AUPR Out
BBP	30.0K	$9.5\pm0.2$	$9.0\pm1.4$	$56.8\pm0.9$	$52.1\pm0.7$	$45.3\pm1.3$	$1.3\pm0.1$
MC-dropout	14.9K	$8.8\pm0.0$	$6.1\pm0.5$	$53.0\pm1.2$	$53.7\pm0.6$	$99.2\pm0.2$	$1.1\pm0.1$
ANPs-SDE-Net	288.5K	$8.8\pm0.0$	$44.4\pm4.2$	$84.2\pm1.6$	$75.2\pm1.7$	$99.8\pm0.0$	$28.8\pm2.0$
CNPs-SDE-Net	141.0K	$8.9\pm0.1$	$7.9 \pm 1.2$	$59.3 \pm 1.5$	$58.6\pm0.9$	$99.2\pm0.1$	$1.3\pm0.1$
SDE-Net	12.4K	$8.7\pm0.1$	$64.3\pm0.6$	$84.1\pm1.1$	$80.6\pm0.5$	$99.7\pm0.0$	$24.7\pm1.0$

4.2.3. ConvCNPs-SDE-Net for Image Classification Dataset: MNIST

The benchmark MNIST dataset is a dataset of handwritten digits from 0 to 9, which consists of 70,000  $28 \times 28$  monochrome images, including 60,000 training images and 10,000 test images [38]. The results of Figure 7 show that even when 70% of the original MNIST dataset is lost, we can roughly distinguish the values of the completed digits with the naked eye.

The drift net in SDE-Net can precisely fit the ID dataset MNIST for classification, and the diffusion net of SDE-Net directly models the relationship between the ID dataset MNIST and epistemic uncertainty; this idea encourages SDE-Net to output greater uncertainty for OOD dataset SVHH and low uncertainty for ID dataset MNIST. However, SDE-Net needs to improve the test performance for receiving the ID dataset MNIST with MR. Thus, we evaluated the performance of the BBP, MC-dropout, ConvCNPs–SDE, and SDE-Net models for OOD detection and ID with MR in classification tasks. MR takes values from [0.1, 0.3, 0.5, 0.7, 0.9, RMR], where RMR denotes random sampling from [0.5, 0.7, 0.9] each time. We report the average performance and standard deviation for five random initializations.



**Figure 7.** ConvCNPs-Net for MNIST with MR. (**a**–**d**) show that the missing rate takes values from [0.1, 0.3, 0.7, 0.9], respectively, to test the performance of the trained ConvCNPs-Net and ConvCNPs–SDE models. For each panel, the top row shows the original ID data MNIST, the second row demonstrates the mask based on MR, the third row exhibits the ID MNIST data with the mask, and the bottom row displays how the masked MNIST data are completed by the ConvCNPs of the ConvCNP–SDE model.

Table 3 shows that as the MR increases, the ConvCNPs–SDE model gradually exceeds the vanilla SDE-Net, BBP, and MC-dropout models in all metrics, including the MR obtained by random sampling.

Aside from OOD data detection, it is also significant that the application of uncertainty makes the model aware of the possibility of making mistakes in test time. Hence, the misclassification detection aims at exploiting the predictive uncertainty to distinguish the test dataset with MR in which the model has misclassified [19,30]. Table 4 shows the misclassification detection performance of the BBP, MC-dropout, ConvCNPs–SDE, and SDE-Net models on ID dataset MNIST with MR = [0.1, 0.3, 0.5, 0.7, 0.9, RMR] and SVHN. We report the average performance and standard deviation for five random initializations.

Table 4 shows that as the MR increases, the ConvCNPs–SDE model consistently surpasses the BBP, MC-dropout, and vanilla SDE-Net models in the first four metrics, including the MR obtained by random sampling. Possibly as a result of the imbalance of the test ID and OOD datasets, BBP achieves comparable or even better performance compared to the MC-dropout, ConvCNPs–SDE, and SDE-Net models in the last metric *AUPR err*. Overall, the ConvCNPs–SDE model is a better model for misclassification tasks in practice.

## 4.2.4. ConvCNPs-SDE-Net for Image Classification Dataset: CIFAR10

The benchmark dataset CIFAR10 has 60,000 color images with 10 classes, including 50,000 training images and 10,000 test images; each class has 6000 images, and the pixel of each image is  $3 \times 32 \times 32$  [39].

MNIST with MR	Model	Classification Accuracy	TNR at TPR 95%	AUROC	Detection Accuracy	AUPR In	AUPR Out
MR = 0.1	BBP MC-dropout ConvCNPs-SDE-Net SDE-Net	$\begin{array}{c} 88.76 \pm 1.73 \\ 98.90 \pm 0.06 \\ 99.32 \pm 0.06 \\ 98.87 \pm 0.06 \end{array}$	$\begin{array}{c} 33.17 \pm 5.09 \\ 88.66 \pm 0.04 \\ 98.69 \pm 0.04 \\ 99.40 \pm 0.03 \end{array}$	$\begin{array}{c} 87.77 \pm 2.03 \\ 96.22 \pm 0.04 \\ 99.68 \pm 0.00 \\ 99.86 \pm 0.01 \end{array}$	$\begin{array}{c} 83.75 \pm 2.32 \\ 92.18 \pm 0.02 \\ 97.99 \pm 0.02 \\ 98.82 \pm 0.04 \end{array}$	$\begin{array}{c} 85.85 \pm 3.21 \\ 89.23 \pm 0.05 \\ 99.01 \pm 0.01 \\ 99.61 \pm 0.02 \end{array}$	$\begin{array}{c} 94.02 \pm 3.42 \\ 98.44 \pm 0.03 \\ 99.89 \pm 0.00 \\ 99.94 \pm 0.01 \end{array}$
MR = 0.3	BBP MC-dropout ConvCNPs-SDE-Net SDE-Net	$\begin{array}{c} 78.37 \pm 3.10 \\ 93.47 \pm 0.06 \\ 98.94 \pm 0.07 \\ 94.98 \pm 0.19 \end{array}$	$\begin{array}{c} 24.90 \pm 3.12 \\ 50.10 \pm 0.06 \\ 99.04 \pm 0.04 \\ 99.47 \pm 0.02 \end{array}$	$\begin{array}{c} 82.06 \pm 4.12 \\ 92.01 \pm 0.03 \\ 99.78 \pm 0.01 \\ 99.70 \pm 0.03 \end{array}$	$\begin{array}{c} 74.46 \pm 2.27 \\ 86.84 \pm 0.06 \\ 98.27 \pm 0.06 \\ 98.96 \pm 0.03 \end{array}$	$\begin{array}{c} 73.86 \pm 2.15 \\ 84.68 \pm 0.03 \\ 99.34 \pm 0.03 \\ 99.51 \pm 0.06 \end{array}$	$\begin{array}{c} 90.22 \pm 2.30 \\ 95.75 \pm 0.05 \\ 99.92 \pm 0.01 \\ 99.82 \pm 0.04 \end{array}$
MR = 0.5	BBP MC-dropout ConvCNPs-SDE-Net SDE-Net	$\begin{array}{c} 52.50 \pm 4.15 \\ 75.88 \pm 0.10 \\ 97.45 \pm 0.17 \\ 80.54 \pm 0.36 \end{array}$	$\begin{array}{c} 16.70 \pm 2.10 \\ 21.72 \pm 0.09 \\ 98.58 \pm 0.03 \\ 99.17 \pm 0.05 \end{array}$	$\begin{array}{c} 70.47 \pm 3.12 \\ 81.64 \pm 0.09 \\ 99.70 \pm 0.01 \\ 98.45 \pm 0.08 \end{array}$	$\begin{array}{c} 78.37 \pm 3.10 \\ 75.87 \pm 0.06 \\ 98.15 \pm 0.05 \\ 97.26 \pm 0.04 \end{array}$	$\begin{array}{c} 54.43 \pm 3.34 \\ 70.73 \pm 0.04 \\ 99.17 \pm 0.02 \\ 98.06 \pm 0.05 \end{array}$	$\begin{array}{c} 84.60 \pm 2.12 \\ 89.60 \pm 0.05 \\ 99.87 \pm 0.01 \\ 98.96 \pm 0.07 \end{array}$
MR = 0.7	BBP MC-dropout ConvCNPs-SDE-Net SDE-Net	$\begin{array}{c} 23.71 \pm 4.11 \\ 46.48 \pm 0.10 \\ 88.96 \pm 0.33 \\ 49.25 \pm 0.11 \end{array}$	$\begin{array}{c} 17.40 \pm 2.23 \\ 12.16 \pm 0.16 \\ 97.45 \pm 0.06 \\ 45.25 \pm 1.36 \end{array}$	$\begin{array}{c} 66.36 \pm 3.12 \\ 70.32 \pm 0.11 \\ 99.20 \pm 0.03 \\ 93.53 \pm 0.13 \end{array}$	$\begin{array}{c} 61.91 \pm 3.23 \\ 65.67 \pm 0.16 \\ 97.47 \pm 0.07 \\ 90.68 \pm 0.15 \end{array}$	$\begin{array}{c} 43.01 \pm 2.44 \\ 53.76 \pm 0.21 \\ 98.03 \pm 0.04 \\ 92.36 \pm 0.16 \end{array}$	$\begin{array}{c} 83.18 \pm 2.23 \\ 83.06 \pm 0.15 \\ 99.62 \pm 0.03 \\ 95.68 \pm 0.12 \end{array}$
MR = 0.9	BBP MC-dropout ConvCNPs-SDE-Net SDE-Net	$\begin{array}{c} 10.08 \pm 1.14 \\ 17.15 \pm 0.54 \\ 36.54 \pm 0.58 \\ 14.56 \pm 0.34 \end{array}$	$\begin{array}{c} 54.34 \pm 2.34 \\ 8.41 \pm 0.44 \\ 79.72 \pm 0.65 \\ 10.30 \pm 0.49 \end{array}$	$\begin{array}{c} 84.13 \pm 4.34 \\ 60.38 \pm 0.35 \\ 96.20 \pm 0.06 \\ 71.82 \pm 0.25 \end{array}$	$\begin{array}{c} 77.78 \pm 1.44 \\ 57.66 \pm 0.34 \\ 93.82 \pm 0.14 \\ 67.83 \pm 0.18 \end{array}$	$\begin{array}{c} 59.63 \pm 2.33 \\ 38.11 \pm 0.24 \\ 92.86 \pm 0.16 \\ 64.34 \pm 0.30 \end{array}$	$\begin{array}{c} 93.72 \pm 1.34 \\ 77.82 \pm 0.36 \\ 97.81 \pm 0.08 \\ 82.80 \pm 0.23 \end{array}$
MR = RMR	BBP MC-dropout ConvCNPs-SDE-Net SDE-Net	$\begin{array}{c} 36.90 \pm 10.56 \\ 42.97 \pm 1.60 \\ 68.90 \pm 11.66 \\ 45.34 \pm 13.66 \end{array}$	$\begin{array}{c} 26.37 \pm 3.63 \\ 11.26 \pm 2.41 \\ 95.72 \pm 1.60 \\ 22.73 \pm 7.30 \end{array}$	$\begin{array}{c} 73.05 \pm 2.26 \\ 69.34 \pm 1.43 \\ 97.89 \pm 0.85 \\ 86.24 \pm 5.13 \end{array}$	$\begin{array}{c} 67.85 \pm 1.62 \\ 64.95 \pm 1.20 \\ 95.91 \pm 0.96 \\ 83.25 \pm 5.82 \end{array}$	$\begin{array}{c} 51.52 \pm 2.56 \\ 53.56 \pm 2.62 \\ 95.70 \pm 1.67 \\ 83.74 \pm 6.71 \end{array}$	$\begin{array}{c} 87.43 \pm 1.22 \\ 82.43 \pm 1.12 \\ 98.89 \pm 0.45 \\ 91.00 \pm 3.11 \end{array}$

**Table 3.** OOD detection results of the BBP, MC-dropout, ConvCNPs–SDE, and SDE-Net models on ID dataset MNIST with MR = [0.1, 0.3, 0.5, 0.7, 0.9, RMR] and OOD dataset SVHN.

**Table 4.** Misclassification detection performance of the BBP, MC-dropout, ConvCNPs–SDE, and SDE-Net models on MNIST with MR = [0.1, 0.3, 0.5, 0.7, 0.9, RMR] and SVHN.

MNIST with MR	Model	TNR at TPR 95%	AUROC	Detection Accuracy	AUPR Succ	AUPR Err
MR = 0.1	BBP MC-dropout ConvCNPs-SDE-Net	$40.78 \pm 2.34$ $86.52 \pm 1.22$ $92.44 \pm 1.06$	$89.55 \pm 0.92$ $95.59 \pm 0.88$ $98.15 \pm 0.12$	$82.52 \pm 1.25$ $91.76 \pm 0.68$ $95.01 \pm 0.50$	$98.79 \pm 0.22 \\99.93 \pm 0.01 \\99.99 \pm 0.00$	$44.14 \pm 3.22 \\ 36.68 \pm 1.88 \\ 31.49 \pm 5.90$
	SDE-Net	$84.43 \pm 3.22$	$96.75 \pm 0.60$	$92.59 \pm 0.76$	$99.96 \pm 0.01$	$31.32 \pm 2.89$
MR = 0.3	BBP MC-dropout ConvCNPs-SDE-Net SDE-Net	$\begin{array}{c} 24.90 \pm 2.26 \\ 60.92 \pm 1.24 \\ 85.80 \pm 3.63 \\ 54.25 \pm 1.51 \end{array}$	$\begin{array}{c} 82.15 \pm 2.82 \\ 91.73 \pm 0.41 \\ 97.22 \pm 0.42 \\ 91.11 \pm 0.46 \end{array}$	$\begin{array}{c} 75.45 \pm 1.32 \\ 54.25 \pm 1.51 \\ 93.00 \pm 1.02 \\ 84.01 \pm 0.93 \end{array}$	$\begin{array}{c} 94.61 \pm 2.34 \\ 99.21 \pm 0.01 \\ 99.97 \pm 0.01 \\ 99.44 \pm 0.04 \end{array}$	$\begin{array}{c} 54.26 \pm 2.04 \\ 46.33 \pm 1.31 \\ 35.81 \pm 4.71 \\ 40.00 \pm 1.75 \end{array}$
MR = 0.5	BBP MC-dropout ConvCNPs-SDE-Net SDE-Net	$\begin{array}{c} 12.17 \pm 2.16 \\ 27.28 \pm 0.33 \\ 72.53 \pm 3.26 \\ 33.07 \pm 0.53 \end{array}$	$\begin{array}{c} 72.48 \pm 1.26 \\ 82.15 \pm 0.33 \\ 95.07 \pm 0.51 \\ 83.79 \pm 0.35 \end{array}$	$\begin{array}{c} 68.08 \pm 1.52 \\ 76.10 \pm 0.23 \\ 89.36 \pm 0.39 \\ 76.12 \pm 0.24 \end{array}$	$\begin{array}{c} 78.42 \pm 4.25 \\ 92.85 \pm 0.13 \\ 99.86 \pm 0.01 \\ 95.33 \pm 0.14 \end{array}$	$\begin{array}{c} 68.32 \pm 2.55 \\ 56.85 \pm 0.43 \\ 36.57 \pm 4.60 \\ 56.45 \pm 0.83 \end{array}$
MR = 0.7	BBP MC-dropout ConvCNPs-SDE-Net SDE-Net	$\begin{array}{c} 10.17 \pm 1.27 \\ 15.67 \pm 0.87 \\ 43.58 \pm 1.67 \\ 28.88 \pm 0.97 \end{array}$	$\begin{array}{c} 65.10 \pm 1.21 \\ 73.82 \pm 0.35 \\ 89.10 \pm 0.16 \\ 76.12 \pm 0.50 \end{array}$	$\begin{array}{c} 65.09 \pm 0.82 \\ 68.55 \pm 0.32 \\ 82.10 \pm 0.41 \\ 68.84 \pm 0.47 \end{array}$	$\begin{array}{c} 51.87 \pm 0.57 \\ 72.23 \pm 0.37 \\ 98.44 \pm 0.07 \\ 75.88 \pm 0.34 \end{array}$	$\begin{array}{c} 82.44 \pm 1.22 \\ 72.72 \pm 0.33 \\ 48.40 \pm 1.01 \\ 75.61 \pm 0.76 \end{array}$
MR = 0.9	BBP MC-dropout ConvCNPs-SDE-Net SDE-Net	$\begin{array}{c} 9.21 \pm 1.22 \\ 7.45 \pm 0.32 \\ 15.73 \pm 0.67 \\ 7.14 \pm 0.42 \end{array}$	$\begin{array}{c} 69.39 \pm 1.45 \\ 59.26 \pm 0.43 \\ 70.02 \pm 0.70 \\ 60.91 \pm 0.41 \end{array}$	$\begin{array}{c} 64.41 \pm 0.62 \\ 57.29 \pm 0.12 \\ 64.73 \pm 0.81 \\ 59.25 \pm 0.30 \end{array}$	$\begin{array}{c} 26.28 \pm 1.46 \\ 25.06 \pm 0.22 \\ 60.84 \pm 1.19 \\ 22.04 \pm 0.79 \end{array}$	$\begin{array}{c} 94.27 \pm 0.32 \\ 86.16 \pm 0.22 \\ 77.94 \pm 0.59 \\ 88.79 \pm 0.25 \end{array}$
MR = RMR	BBP MC-dropout ConvCNPs-SDE-Net SDE-Net	$\begin{array}{c} 10.85 \pm 2.43 \\ 34.81 \pm 13.32 \\ 36.95 \pm 11.08 \\ 35.11 \pm 15.40 \end{array}$	$\begin{array}{c} 70.43 \pm 5.52 \\ 78.00 \pm 5.40 \\ 88.19 \pm 2.74 \\ 83.44 \pm 6.02 \end{array}$	$\begin{array}{c} 67.83 \pm 3.42 \\ 72.60 \pm 4.40 \\ 81.51 \pm 2.59 \\ 76.25 \pm 4.80 \end{array}$	$\begin{array}{c} 68.10 \pm 6.22 \\ 77.47 \pm 9.45 \\ 94.38 \pm 3.33 \\ 80.15 \pm 15.44 \end{array}$	$\begin{array}{c} 80.86 \pm 3.23 \\ 76.60 \pm 4.21 \\ 71.57 \pm 6.18 \\ 83.92 \pm 3.18 \end{array}$

Figure 8 shows that even when 70% of the original CIFAR10 is lost, we can generally distinguish the objects in the completed images with the naked eye in the bottom row. We evaluate the performance of the ConvCNPs–SDE and SDE-Net models for OOD dataset SVHN detection and ID dataset CIFAR10 with MR in classification tasks. We report the average performance and standard deviation for five random initializations.



**Figure 8.** ConvCNPs-Net for CIFAR10 with MR. (**a**–**d**) show that the missing rate takes values from [0.1, 0.3, 0.7, 0.9], respectively, to test the performance of the trained ConvCNPs-Net and ConvCNP–SDE models. For each panel, the top row shows the original ID data of CIFAR10, the second row demonstrates the mask based on MR, the third row exhibits the CIFAR10 data with the mask, and the bottom row displays how the masked CIFAR10 data are completed by the ConvCNPs of the ConvCNP–SDE model.

Table 5 shows that the ConvCNPs–SDE model exceeds the other models in the important classification accuracy metric, while BBP surpasses almost all of the other models in the remaining four metrics for all MR values, which shows that the ability of BBP to identify SVHN as OOD data is better than that of the other methods. From the perspective of classification accuracy, the ConvCNPs–SDE model is a better model for OOD detection with benchmark dataset CIFAR10 in practice.

Table 6 shows the misclassification detection performance of the BBP, MC-dropout, ConvCNPs–SDE, and SDE-Net models on ID dataset CIFAR10 with MR and SVHN. We report the average performance and standard deviation for five random initializations. For the different values of MR, the ConvCNPs–SDE model surpasses the BBP, MC-dropout, and vanilla SDE-Net models in the first four metrics. Possibly as a result of the imbalance of the test ID and OOD datasets, SDE-Net achieves better performance than the other models in the last metric *AUPR err*. Overall, the ConvCNPs–SDE model is a better model for misclassification tasks in practice for CIFAR10.

**Table 5.** Classification and OOD detection results of the BBP, MC-dropout, ConvCNPs–SDE, and SDE-Net models on ID dataset CIFAR10 with MR = [0.1, 0.3, 0.5, 0.7, 0.9, RMR] and OOD dataset SVHN.

CIFAR10 with MR	Model	Classification Accuracy	TNR at TPR 95%	AUROC	Detection Accuracy	AUPR In	AUPR Out
	BBP	$19.83\pm0.45$	$64.31 \pm 3.35$	$92.57\pm2.47$	$86.49 \pm 2.45$	$84.29 \pm 1.55$	$96.46 \pm 1.42$
<b>MD</b> 0.1	MC-dropout	$43.01\pm0.35$	$3.67\pm0.13$	$50.90 \pm 0.15$	$52.50\pm0.10$	$31.15\pm0.16$	$71.50\pm0.05$
MR = 0.1	ConvCNPs-SDE-Net	$78.66\pm0.20$	$4.46\pm0.31$	$59.97 \pm 0.16$	$59.33 \pm 0.11$	$42.23\pm0.15$	$75.54\pm0.07$
	SDE-Net	$23.23\pm0.25$	$1.57\pm0.10$	$37.79\pm0.13$	$50.36\pm0.06$	$24.04\pm0.14$	$63.41\pm0.06$
	BBP	$19.64\pm0.15$	$59.53 \pm 3.43$	$92.62 \pm 1.89$	$86.50 \pm 1.55$	$86.45 \pm 2.45$	$96.36 \pm 1.45$
MP = 0.2	MC-dropout	$24.91\pm0.25$	$1.87\pm0.15$	$40.81\pm0.35$	$50.01\pm0.26$	$23.62\pm0.12$	$65.83 \pm 0.25$
MIK = 0.3	ConvCNPs-SDE-Net	$76.46\pm0.20$	$4.25\pm0.37$	$58.31 \pm 0.30$	$57.89 \pm 0.20$	$40.55\pm0.18$	$74.71\pm0.22$
	SDE-Net	$13.11\pm2.89$	$4.61\pm0.25$	$52.28\pm0.34$	$52.55\pm0.15$	$31.46\pm0.12$	$72.74\pm0.30$
	BBP	$19.44\pm0.25$	$46.25\pm4.26$	$87.53\pm3.72$	$81.05\pm3.72$	$77.70\pm3.38$	$93.95 \pm 1.68$
MP = 0F	MC-dropout	$18.49\pm0.19$	$2.22\pm0.15$	$40.79\pm0.19$	$50.00\pm0.15$	$23.42\pm0.17$	$66.10\pm0.16$
MIK = 0.5	ConvCNPs-SDE-Net	$72.21\pm0.26$	$3.57\pm0.19$	$55.08 \pm 0.28$	$55.35\pm0.13$	$37.36\pm0.15$	$72.98 \pm 0.22$
	SDE-Net	$10.33\pm0.06$	$4.97\pm0.08$	$50.74\pm0.26$	$50.95\pm0.17$	$29.18\pm0.21$	$72.33\pm0.13$
	BBP	$18.39\pm0.52$	$21.77 \pm 4.09$	$77.28 \pm 3.11$	$70.66 \pm 2.47$	$67.15 \pm 2.63$	$87.66 \pm 2.29$
MP = 0.7	MC-dropout	$14.61\pm0.18$	$2.76\pm0.28$	$42.66\pm0.22$	$50.00\pm0.13$	$24.26\pm0.23$	$67.36\pm0.14$
MIK = 0.7	ConvCNPs-SDE-Net	$62.47\pm0.49$	$2.50\pm0.06$	$48.70\pm0.28$	$52.08\pm0.05$	$31.64\pm0.14$	$69.60\pm0.13$
	SDE-Net	$10.10\pm0.08$	$4.91\pm0.28$	$49.86\pm0.50$	$50.27\pm0.19$	$27.85\pm0.31$	$72.05\pm0.34$
	BBP	$14.60\pm0.67$	$1.10\pm0.40$	$55.31 \pm 2.84$	$65.55 \pm 1.23$	$54.23 \pm 2.43$	$68.97 \pm 0.93$
MP = 0.0	MC-dropout	$12.01\pm0.16$	$2.81\pm0.11$	$43.53\pm0.06$	$50.00\pm0.07$	$24.45\pm0.04$	$67.91\pm0.16$
MIK = 0.9	ConvCNPs-SDE-Net	$36.48\pm0.43$	$1.29\pm0.01$	$37.86\pm0.39$	$50.49 \pm 0.05$	$23.48\pm0.30$	$64.32\pm0.18$
	SDE-Net	$10.18\pm0.06$	$4.91\pm0.18$	$49.40\pm0.06$	$50.19\pm0.12$	$27.02\pm0.05$	$71.92\pm0.11$
	BBP	$16.37\pm0.38$	$15.04\pm5.27$	$70.19 \pm 1.22$	$65.72\pm0.64$	$59.93 \pm 0.21$	$83.36 \pm 1.87$
	MC-dropout	$14.78\pm0.08$	$2.31\pm0.25$	$41.87\pm0.35$	$50.00\pm0.15$	$23.72\pm0.23$	$66.80\pm0.33$
WIK = KWIK	ConvCNPs-SDE-Net	$56.04 \pm 4.08$	$1.98\pm0.24$	$46.79\pm2.04$	$51.80\pm0.48$	$30.41 \pm 1.90$	$68.45\pm0.98$
	SDE-Net	$10.18\pm0.05$	$4.82\pm0.26$	$50.01\pm0.36$	$50.42\pm0.22$	$28.00\pm0.28$	$72.03\pm0.27$

**Table 6.** Misclassification detection performance of the BBP, MC-dropout, ConvCNPs-SDE, and SDE-Net models on CIFAR10 with MR = [0.1, 0.3, 0.5, 0.7, 0.9, RMR] and SVHN.

CIFAR10 with MR	Model	TNR at TPR 95%	AUROC	Detection Accuracy	AUPR Succ	AUPR Err
	BBP	$10.97\pm0.66$	$58.75\pm0.62$	$56.19 \pm 0.70$	$25.69\pm0.32$	$85.16\pm0.77$
<b>MD</b> 0.1	MC-dropout	$10.89\pm0.60$	$67.19\pm0.12$	$63.17\pm0.23$	$61.69\pm0.13$	$69.60\pm0.24$
MK = 0.1	ConvCNPs-SDE-Net	$27.47\pm0.80$	$83.77\pm0.18$	$77.33\pm0.26$	$94.88\pm0.12$	$54.12\pm0.64$
	SDE-Net	$21.85 \pm 1.60$	$72.51\pm0.29$	$66.64 \pm 0.35$	$44.85\pm0.58$	$88.73\pm0.21$
	BBP	$9.84\pm0.34$	$56.17\pm0.24$	$54.57\pm0.34$	$24.77\pm0.14$	$84.32\pm0.67$
MR = 0.3	MC-dropout	$6.47\pm0.34$	$58.22\pm0.26$	$56.90 \pm 0.46$	$32.83\pm0.66$	$78.92\pm0.24$
	ConvCNPs-SDE-Net	$26.96\pm0.83$	$82.70\pm0.24$	$76.04\pm0.37$	$93.83\pm0.17$	$55.70\pm0.50$
	SDE-Net	$8.61 \pm 1.24$	$65.01\pm0.68$	$63.04\pm0.57$	$22.85\pm0.88$	$91.82\pm0.34$
	BBP	$8.19\pm0.02$	$55.73 \pm 0.68$	$55.17\pm0.70$	$23.32\pm0.43$	$83.74\pm0.02$
	MC-dropout	$6.11\pm0.14$	$57.18 \pm 0.23$	$56.69 \pm 0.22$	$24.32\pm0.26$	$84.06\pm0.34$
MR = 0.5	ConvCNPs-SDE-Net	$23.55\pm0.79$	$80.31\pm0.16$	$73.78\pm0.33$	$91.32\pm0.05$	$57.15\pm0.57$
	SDE-Net	$5.89 \pm 1.22$	$53.64 \pm 0.83$	$53.40\pm0.19$	$13.31\pm0.32$	$90.45\pm0.33$
	BBP	$9.00\pm0.09$	$53.62\pm0.66$	$53.73\pm0.61$	$20.29 \pm 1.04$	$84.15\pm0.25$
MR = 0.7	MC-dropout	$5.89 \pm 0.78$	$56.51 \pm 0.58$	$56.08 \pm 0.38$	$18.90\pm0.22$	$87.23 \pm 0.38$
	ConvCNPs-SDE-Net	$18.24\pm0.69$	$75.85\pm0.39$	$69.81 \pm 0.40$	$84.37\pm0.38$	$61.18 \pm 1.22$
	SDE-Net	$5.05\pm0.98$	$50.28 \pm 0.51$	$51.17\pm0.40$	$10.67\pm0.44$	$89.87\pm0.28$

CIFAR10 with MR	Model	TNR at TPR 95%	AUROC	Detection Accuracy	AUPR Succ	AUPR Err
MR = 0.9	BBP MC-dropout ConvCNPs-SDE-Net SDE-Net	$\begin{array}{c} 5.06 \pm 0.13 \\ 5.73 \pm 0.78 \\ 11.53 \pm 0.48 \\ 4.81 \pm 0.88 \end{array}$	$\begin{array}{c} 53.45 \pm 2.20 \\ 55.93 \pm 0.35 \\ 66.46 \pm 0.45 \\ 50.93 \pm 0.40 \end{array}$	$\begin{array}{c} 53.78 \pm 1.96 \\ 55.23 \pm 0.28 \\ 62.43 \pm 0.48 \\ 51.68 \pm 0.36 \end{array}$	$\begin{array}{c} 16.89 \pm 1.98 \\ 14.60 \pm 0.75 \\ 56.21 \pm 0.53 \\ 11.26 \pm 0.68 \end{array}$	$\begin{array}{c} 86.25 \pm 0.14 \\ 89.49 \pm 0.23 \\ 74.77 \pm 0.72 \\ 89.86 \pm 0.25 \end{array}$
MR = RMR	BBP MC-dropout ConvCNPs-SDE-Net SDE-Net	$\begin{array}{c} 8.12 \pm 0.17 \\ 6.66 \pm 0.58 \\ 16.33 \pm 1.57 \\ 5.34 \pm 0.77 \end{array}$	$\begin{array}{c} 55.30 \pm 1.18 \\ 56.51 \pm 0.88 \\ 75.42 \pm 1.58 \\ 51.45 \pm 0.94 \end{array}$	$\begin{array}{c} 53.92 \pm 0.85 \\ 55.59 \pm 0.83 \\ 69.77 \pm 1.39 \\ 51.74 \pm 0.89 \end{array}$	$\begin{array}{c} 19.66 \pm 0.15 \\ 19.04 \pm 0.55 \\ 80.73 \pm 3.76 \\ 11.69 \pm 0.56 \end{array}$	$\begin{array}{c} 86.11 \pm 0.74 \\ 87.32 \pm 0.25 \\ 66.10 \pm 2.00 \\ 90.21 \pm 0.24 \end{array}$

Table 6. Cont.

#### 5. Discussion

In this work, we proposed to incorporate the NPs family into SDE-Net to form a vNPs–SDE model for handling noisy ID datasets. The vNPs–SDE model was implemented with ConvCNPs-Net for synthetic 1D regression and 2D image classification tasks, and the vNPs–SDE model was implemented with CNPs and ANPs for multidimensional regression tasks.

For the multidimension regression tasks, the results of five models including BBP, MC-dropout, SDE-Net, CNPs–SDE, and ANPs–SDE are demonstrated in Tables 1 and 2. Table 1 shows that as the MR increases, the values of RMSE gradually increase, and the CNPs–SDE model is more accurate than the other models. Table 2 shows that SDE-Net still has the optimal performance in RMSE, TNR at TPR 95%, and detection accuracy. Additionally, compared to the other models, SDE-Net has the fewest parameters. However, ANPs–SDE-Net is a more effective method than the other methods in terms of *AUROC*, *AUPR in*, and *AUPR out*, and is comparable to SDE-Net in RMSE.

The results of the three models—namely, BBP, SDE-Net, and ConvCNPs–SDE—are plotted in Figure 6 for synthetic 1D data. Due to MC-dropout method not fitting the GPs dataset, the results of MC-dropout are not shown in Figure 6. More specifically, our proposed ConvCNPs–SDE model can fit the synthetic 1D data better and produce smaller variances than the BBP and SDE-Net models. When MR is 0.9, we find that the results of the BBP and SDE-Net models deviate from the data. At the beginning and the end of the training set, BBP produces worse results than SDE-Net and ConvCNPs–SDE; this may be due to the learning Bayesian phase at the beginning and the uncertainty introduced about unseen data [8].

The results of the four models, including BBP, MC-dropout, SDE-Net, and ConvCNPs– SDE, are given in Tables 3 and 4 for MNIST. Table 3 shows that as the MR increases, the ConvCNPs–SDE model gradually surpasses the vanilla SDE-Net, BBP, and MC-dropout models in all metrics, including the MR obtained by random sampling; this indicates that even with noisy ID data, our proposed ConvCNPs–SDE model can still effectively detect OOD data. Table 4 describes the misclassification detection when exploiting the proposed ConvCNPs–SDE model to distinguish between the dataset with MR and the OOD data. Table 4 shows that as the MR increases, the ConvCNPs–SDE model consistently surpasses the BBP, MC-dropout, and vanilla SDE-Net models in the first four metrics. However, BBP achieves comparable or better performance compared to the MC-dropout, ConvCNPs–SDE, and SDE-Net modesl in the last metric *AUPR err;* this may be as a result of the imbalance of the ID and OOD data [19], or due to noisy ID data, and as such deserves further study. For the benchmark CIFAR10 dataset, the results are depicted in Tables 5 and 6. Table 5 shows that the ConvCNPs–SDE model is superior to the other methods in terms of classification accuracy, while BBP surpasses almost all of the other models in the remaining four metrics for all ID data with MR values, which means that the ability of BBP to identify SVHN as OOD data is better than that of the other methods. Tables 3 and 5 show the OOD detection for MNIST and CIAFR10, respectively, but we have different conclusions for distinguishing between OOD and noisy ID data. We can assume that the BBP method, as illustrated in Table 5, has some advantages over the other approaches in terms of OOD detection, with three large input channels. For the ID dataset, the limitation of our proposed ConvCNPs–SDE model may be that the ordinary DNNs in the ConvCNPs–SDE model cause poor performance. Tables 4 and 6 have smiliar conclusions.

Despite these promising results, future studies should train and test our models on more datasets in order to verify their performance, and at the same time test the OOD detection perforamnce of BBP on other noisy ID datasets. Moreover, the local optimal characteristics of DNNs lead to instability of the predicted results; as such, future research should explore other methods to make the loss values of DNNs close to a constant.

#### 6. Conclusions

SDE-Net is a much simpler and more straightforward method than BNNs for uncertainty estimates in deep neural networks, and it can separate different sources of uncertainties and accurately distinguish between ID and OOD datasets. It is a promising method for equipping NNs with meaningful uncertainties in many safety-critical fields, such as medical diagnoses and self-driving vehicles. However, SDE-Net does not consider the general situation in a wider field—for instance, ID data with noise or high missing rates in practice.

In this paper, we proposed a vNPs–SDE model, which combines SDE-Net with the NPs family in order to deal with the noisy ID dataset for uncertainty estimates. Specifically, we applied the permutation invariance property of CNPs and ANPs for multimensional regression tasks, and the translation equivariance property of ConvCNPs for synthetic 1D regression and 2D image classification tasks. Extensive experimental results of the vNPs–SDE model show that vNPs can not only improve SDE-Net in terms of OOD detection and misclassification detection between ID and OOD datasets, but also allow it to make more efficient and accurate predictions for ID datasets with missing rates than the BBP, MC-dropout, and vanilla SDE-Net models—except for OOD detection for CIAFR10, in which BBP is superior. Hence, the ability of the BBP model in terms of OOD detection deserves further extensive experiments.

Future studies should consider the local optimal problem of deep learning models, which is also one of the sources of uncertainty that generate unstable predictions—we can find a curved path, and the parameters of DNNs on the path can produce near constant loss of deep learning. The DNNs of the proposed vNPs–SDE model are too simple, but we can apply transfer learning methods to replace the simple DNNs with the state-of-the-art ResNets, such as ResNet-18 and ResNet-34, to improve performance.

**Author Contributions:** Conceptualization, Y.W. and S.Y.; methodology, Y.W. and S.Y.; software, Y.W.; validation, Y.W.; formal analysis, Y.W.; investigation, Y.W.; resources, Y.W.; data curation, Y.W.; writing—original draft preparation, Y.W.; writing—review and editing, S.Y.; visualization, Y.W. All authors have read and agreed to the published version of the manuscript.

**Funding:** The work was supported by the National Key R&D Program of China under Grant No.2018YFB1402600.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Datasets are open access and available on References [33,41,42].

**Acknowledgments:** The authors gratefully acknowledge Li DanDan and Tan HuoBin for their advice and support in reviewing and many other aspects of this work.

**Conflicts of Interest:** The authors declare no conflict of interest.

### Abbreviations

The followin	g abbreviations are used in this manuscript:
SDE-Net	Neural stochastic differential equation model
DNNs	Deep neural networks
ID	In-distribution
OOD	Out-of-distribution
NPs	Neural processes
vNPs	Vanilla neural processes or neural process variants
ConvCNP	Convolutional conditional neural process
CNPs	Conditional neural processes
ANPs	Attentive neural processes
BNNs	Bayesian neural processes
PCA	Principal component analysis
GPs	Gaussian processes
MLP	Multilayer perceptron
CNNs	Convolutional neural networks
ODE-Net	Neural ordinary differential equation
Conv2d	Two-dimensional convolution
RKHS	Reproducing kernel Hilbert space
ResNet	Residual networks
ELBO	Evidence lower bound
KL	Kullback–Leibler divergence

# References

- Krizhevsky, A.; Sutskever, I.; Hinton, G.E. Imagenet classification with deep convolutional neural networks. In Proceedings of the Advances in Neural Information Processing Systems, Lake Tahoe, NV, USA, 3–8 December 2012; pp. 1097–1105.
- He, K.; Zhang, X.; Ren, S.; Sun, J. Deep Residual Learning for Image Recognition. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, Las Vegas, NV, USA, 26 June–1 July 2016; pp. 770–778.
- Singh, S.P.; Kumar, A.; Darbari, H.; Singh, L.; Jain, S. Machine translation using deep learning: An overview. In Proceedings of the 2017 International Conference on Computer, Communications and Electronics (Comptelix), Jaipur, India, 1–2 July 2017; pp. 162–167.
- Mousavi, S.S.; Schukat, M.; Howley, E. Deep Reinforcement Learning: An Overview. In Proceedings of the SAI Intelligent Systems Conference, London, UK, 3–4 September 2018; pp. 426–440.
- Guo, C.; Pleiss, G.; Sun, Y.; Weinberger, K.Q. On calibration of modern neural networks. In Proceedings of the 34th International Conference on Machine Learning, Sydney, NSW, Australia, 6–11 August 2017; Volume 70, pp. 1321–1330.
- 6. MacKay, C.; David, J. Information Theory, Inference and Learning Algorithms; Cambridge University Press: New York, NY, USA, 2003.
- Kingma, D.P.; Salimans, T.; Welling, M. Variational dropout and the local reparameterization trick. In Proceedings of the 28th International Conference on Neural Information Processing Systems, Montréal, Canada, 7–12 December 2015; Volume 2, pp. 2575–2583.
- 8. Blundell, C.; Cornebise, J.; Kavukcuoglu, K.; Wierstra, D. Weight uncertainty in neural network. In Proceedings of the 32nd International Conference on Machine Learning, Lille, France, 6–11 July 2015; pp. 1613–1622.
- Izmailov, P.; Maddox, W.J.; Kirichenko, P.; Garipov, T.; Vetrov, D.P.; Wilson, A.G. Subspace Inference for Bayesian Deep Learning. In Proceedings of the 35th Conference on Uncertainty in Artificial Intelligence, Tel Aviv, Israel, 22–25 July 2019; pp. 1169–1179.
- Garipov, T.; Izmailov, P.; Podoprikhin, D.; Vetrov, D.P.; Wilson, A.G. Loss Surfaces, Mode Connectivity, and Fast Ensembling of DNNs. In Proceedings of the 32nd Conference on Neural Information Processing Systems, Montréal, Canada, 2–8 December 2018; Volume 31, pp. 8789–8798.
- 11. Wang, Y.; Yao, S.; Xu, T. Incremental Kernel Principal Components Subspace Inference with Nyström Approximation for Bayesian Deep Learning. *IEEE Access* 2021, *9*, 36241–36251. [CrossRef]
- 12. Gal, Y.; Ghahramani, Z. Bayesian Convolutional Neural Networks with Bernoulli Approximate Variational Inference. In Proceedings of the ICLR workshop track, San Juan, Puerto Rico, 2–4 May 2016.
- Lakshminarayanan, B.; Pritzel, A.; Blundell, C. Simple and Scalable Predictive Uncertainty Estimation using Deep Ensembles. In Proceedings of the Advances in Neural Information Processing Systems, Long Beach, CA, USA, 3–9 December 2017; Volume 30, pp. 6402–6413.

- Izmailov, P.; Podoprikhin, D.; Garipov, T.; Vetrov, D.P.; Wilson, A.G. Averaging Weights Leads to Wider Optima and Better Generalization. In Proceedings of the 34th Conference on Uncertainty in Artificial Intelligence, Monterey, CA, USA, 6–8 August 2018; pp. 876–885.
- 15. Geifman, Y.; Uziel, G.; El-Yaniv, R. Bias-Reduced Uncertainty Estimation for Deep Neural Classifiers. In Proceedings of the International Conference on Learning Representations, Vancouver, BC, Canada, 30 April–3 May 2018.
- Gal, Y.; Ghahramani, Z. Dropout as a Bayesian approximation: Representing model uncertainty in deep learning. In Proceedings of the 33rd International Conference on International Conference on Machine Learning, New York, NY, USA, 19–24 June 2016; Volume 48, pp. 1050–1059.
- Kendall, A.; Gal, Y. What uncertainties do we need in Bayesian deep learning for computer vision. In Proceedings of the 31st International Conference on Neural Information Processing Systems, Long Beach, CA, USA, 3–9 December 2017; Volume 30, pp. 5580–5590.
- Chen, R.T.Q.; Rubanova, Y.; Bettencourt, J.; Duvenaud, D. Neural ordinary differential equations. In Proceedings of the 32nd International Conference on Neural Information Processing Systems, Montréal, Canada, 2–8 December 2018; Volume 31, pp. 6572–6583.
- 19. Kong, L.; Sun, J.; Zhang, C. SDE-Net: Equipping Deep Neural Networks with Uncertainty Estimates. In Proceedings of the 37th International Conference on Machine Learning, Virtual Event, Vienna, Austria, 13–18 July 2020; Volume 1, pp. 5405–5415.
- 20. Øksendal, B. Stochastic differential equations. In Stochastic Differential Equations; Springer: Berlin, Germany, 2003; p. 11.
- 21. Bass, R.F. Stochastic Processes. In *Cambridge Series in Statistical and Probabilistic Mathematics*; Cambridge University Press: New York, NY, USA, 2011.
- Jeanblanc, M.; Yor, M.; Chesney, M. Continuous-Path Random Processes:Mathematical Prerequisites. In Mathematical Methods for Financial Markets; Avellaneda, M., Barone-Adesi, G., Eds.; Springer: Dordrecht, The Netherlands; Heidelberg, Germany; London, UK; New York, NY, USA, 2009.
- Hendrycks, D.; Gimpel, K. A Baseline for Detecting Misclassified and Out-of-Distribution Examples in Neural Networks. In Proceedings of the international Conference on Learning Representations, Caribe Hilton, San Juan, Puerto Rico, 2–4 May 2016.
- 24. Lee, K.; Lee, H.; Lee, K.; Shin, J. Training Confidence-calibrated Classifiers for Detecting Out-of-Distribution Samples. In Proceedings of the International Conference on Learning Representations, Vancouver, BC, Canada, 30 April–3 May 2018.
- Malinin, A.; Gales, M. Predictive uncertainty estimation via prior networks. In Proceedings of the 32nd Conference on Neural Information Processing Systems, Montréal, Canada, 2–8 December 2018; Volume 31, pp. 7047–7058.
- Arpit, D.; Stanisław, J.; Nicolas, B.; David, K.; Emmanuel, B.; Maxinder, S.K.; Tegan, M.; Asja, F.; Aaron, C.; Yoshua, B. A Closer Look at Memorization in Deep Networks. In Proceedings of the 34th International Conference on Machine Learning, Sydney, NSW, Australia, 6–11 August 2017; Volume 70, pp. 233–242.
- Jiang, L.; Huang, D.; Liu, M.; Yang, W. Beyond Synthetic Noise: Deep Learning on Controlled Noisy Labels. In Proceedings of the 37th International Conference on Machine Learning, Vienna, Austria, 12–18 July 2020; Volume 1, pp. 4804–4815.
- 28. Garnelo, M.; Schwarz, J.; Rosenbaum, D.; Viola, F.; Rezende, D.J.; Eslami, S.M.A.; Teh, Y.W. Neural Processes. *arXiv* 2018, arXiv:1807.01622.
- Garnelo, M.; Rosenbaum, D.; Maddison, C.; Ramalho, T.; Saxton, D.; Shanahan, M.; The, Y.W.; Rezende, D.J.; Eslami, A. Conditional neural processes. In Proceedings of the International Conference on Machine Learning, Stockholm, Sweden, 10–15 July 2018; pp. 1690–1699.
- Kim, H.; Mnih, A.; Schwarz, J.; Garnelo, M.; Eslami, S.M.A.; Rosenbaum, D.; Vinyals, O.; The, Y.W. Attentive Neural Processes. In Proceedings of the International Conference on Learning Representations, New Orleans, LA, USA, 6–9 May 2019.
- Gordon, J.; Bruinsma, W.P.; Foong, A.Y.K.; Requeima, J.; Dubois, Y.; Turner, R.E. Convolutional Conditional Neural Processes. In Proceedings of the 8th International Conference on Learning Representations, Formerly Addis Ababa, Ethiopia, 26 April–1 May 2020.
- Vaswani, A.; Shazeer, N.; Parmar, N.; Uszkoreit, J.; Jones, L.; Gomez, A.N.; Kaiser, L.; Polosukhin, I. Attention is all you need. In Proceedings of the 31st International Conference on Neural Information Processing Systems, Long Beach, CA, USA, 4–9 December 2017; Volume 30, pp. 5998–6008.
- LeCun, Y.; Bottou, L.; Bengio, Y.; Haffner, P. Gradient-based learning applied to document recognition. *Proc. IEEE* 1998, *86*, 2278–2324. Available online: http://yann.lecun.com/exdb/mnist/ (accessed on 30 November 1998). [CrossRef]
- Cohen, T.S.; Welling, M. Group equivariant convolutional networks. In Proceedings of the 33rd International Conference on International Conference on Machine Learning, New York, NY, USA, 20–22 June 2016; Volume 48, pp. 2990–2999.
- 35. Aronszajn, N. Theory of Reproducing Kernels. Trans. Am. Math. Soc. 1950, 68, 337–404. [CrossRef]
- 36. Zaheer, M.; Kottur, S.; Ravanbakhsh, S.; Poczos, B.; Salakhutdinov, R.R.; Smola, A.J. Deep Sets. In Proceedings of the Advances in Neural Information Processing Systems, Long Beach, CA, USA, 4 December 2017; Volume 30, pp. 3391–3401.
- 37. He, K.; Zhang, X.; Ren, S.; Sun, J. Identity Mappings in Deep Residual Networks. In Proceedings of the European Conference on Computer Vision, Amsterdam, The Netherlands, 8–16 October 2016; pp. 630–645.
- Rezende, D.; Mohamed, S. Variational Inference with Normalizing Flows. In Proceedings of the 32nd International Conference on Machine Learning, Lile, France, 6–11 July 2015; pp. 1530–1538.
- 39. Raissi, M.; Karniadakis, G.E. Hidden physics models: Machine learning of nonlinear partial differential equations. *J. Comput. Phys.* **2018**, 357, 125–141. [CrossRef]

- 40. Fawcett, T. An introduction to ROC analysis. Pattern Recognit. Lett. 2006, 27, 861–874. [CrossRef]
- 41. Dua, D.; Graff, C. UCI Machine Learning Repository. 2017. Available online: http://archive.ics.uci.edu/ml (accessed on 31 December 2017).
- 42. Krizhevsky, A.; Hinton, G. Learning Multiple Layers of Features from Tiny Images. 2009. Available online: http://www.cs. toronto.edu/~{}kriz/cifar.html (accessed on 8 April 2009).